BUKTI KORESPONDENSI

ARTIKEL JURNAL INTERNASIONAL BEREPUTASI

Judul Artikel : Microfinance and microenterprise performance in Indonesia: an

extended and updated survey

Jurnal : International Journal of Social Economics, 2018, volume 45(6),

957-972

Penulis : Adwin Surja Atmadja, Parmendra Sharma dan Jen-Je Su

No	Perihal	Tanggal
1	Bukti konfirmasi submit artikel dan artikel yang disubmit	1 Peb 2017
2	Bukti korespondensi permintaan revisi oleh reviewer	14 Sep 2017
3	Bukti konfirmasi submit revisi dan artikel yang direvisi	26 Okt 2017
4	Bukti konfirmasi artikel accepted	27 Okt 2017

Bukti konfirmasi submit artikel dan artikel yang disubmit (1 Pebruari 2017)

International Journal of Social Economics - Manuscript ID IJSE-02-2017-0031

International Journal of Social Economics <onbehalfof+j.pearce+2010.hull.ac.uk@manuscriptcentral.com>

Rab 01/02/2017 08.20

Kepada: adwin.atmadja@griffithuni.edu.au < adwin.atmadja@griffithuni.edu.au >; p.sharma@griffith.edu.au >; p.sharma@griffithuni.edu.au >; p.<p.sharma@griffith.edu.au>;j.su@griffith.edu.au <j.su@griffith.edu.au>

01-Feb-2017

Dear Dr. Sharma:

Your manuscript entitled "Microfinance and microenterprise performance in Indonesia: an extended and updated survey" has been successfully submitted online and is presently being given full consideration for publication in the International Journal of Social Economics.

Your manuscript ID is IJSE-02-2017-0031.

Please mention the above manuscript ID in all future correspondence or when calling the office for questions. If there are any changes in your street address or e-mail address, please log in to Manuscript Central at https://mc.manuscriptcentral.com/ijsec and edit your user information as appropriate.

You can also view the status of your manuscript at any time by checking your Author Centre after logging in to https://mc.manuscriptcentral.com/ijsec.

Thank you for submitting your manuscript to the International Journal of Social Economics.

Sincerely,

International Journal of Social Economics Editorial Office

P.S. Don't forget, you can read the journal online at http://www.emeraldinsight.com/0306-8293.htm



International Journal of Social Economics

Microfinance and microenterprise performance in Indonesia: an extended and updated survey

Journal:	International Journal of Social Economics
Manuscript ID	IJSE-02-2017-0031
Manuscript Type:	Research Paper
Keywords:	microfinance, microenterprise performance, money separation

SCHOLARONE™ Manuscripts

Microfinance and microenterprise performance in Indonesia: an extended and updated survey

Structured abstract

Purpose

This study addresses the small, women micro-entrepreneur dominated and heterogeneity limitations of the Atmadja et al. (2016) study. The sample is much larger, includes more men and is more heterogeneous, which allows deeper insights and more meaningful explanation of the relationship between microfinance and microenterprise performance in the case of Indonesia, including the effects of gender, lending scheme and money separation.

Design/methodology/approach

Survey of 556 respondents across five micro-credit providers in the city of Surabaya using updated instrument. Ordered probit is used to analyse data.

Findings

Microfinance may not matter for microenterprise performance in the case of Indonesia. Additionally, microcredit schemes (individual versus group) and gender may also not matter for performance, but money separation might have some influence.

Practical implications

Non-financial factors such as human capital, spousal involvement, and money separation should be considered as important factors for improving microenterprise business performance in Indonesia, with less focus on microcredit per se.

Originality/value

This study provides further evidence that microfinance may not matter for microenterprise performance in the case of Indonesia, a populous middle income country with a very long history of microfinance.

1. Introduction

Debates abound and long-standing on the effects of microfinance on microenterprise performance; see for example Barnes et al. (2001), Chen and Snodgrass (2001), Copestake et al. (2001), Dunn and Arbuckle Jr (2001), Kondo et al. (2008), and Crépon et al. (2011) who argue for a positive relationship, and Adams and Von Pischke (1992), Olson et al. (2003), Davis (2006), and Bateman and Chang (2012) who argue that microfinance may not really matter for performance.

Despite a very long history of microfinance in Indonesia¹—a populous, middle income county bestowed with constant and concerning levels of poverty and economic challenges—little remains known about the effects of microfinance on microenterprise performance in the country. To the best of our knowledge, the most relevant study so far might be by Atmadja et al. (2016) who use data obtained from a 2010 survey to provide an in–depth understanding of the role of microfinance for microenterprise performance in Surabaya, the second largest city in Indonesia. Essentially, the findings suggest that microcredit itself does not necessarily improve business performance.

¹ The Bank Priyayi of Purwokerto established in 1895 has been known as the first formal commercial microfinance institution in Indonesia (BWTP 2013; Holloh 2001).

While the 2010 survey data has also been used in other published works such as Bradley et al. (2011) and Wood et al. (2015), Atmadja et al. (2016) acknowledge a number of limitations of the survey with respect to their specific investigation, i.e. testing the relationship between microfinance and microenterprise performance, and suggest that a larger sample, including men—owned microenterprises and greater heterogeneity might provide deeper insights and more meaningful explanation of the relationship between microfinance and microenterprise performance.

The above has exactly been the purpose of the present study. Essentially, we have gone back to the city of Surabaya and conducted a much wider and larger survey—we have extended and updated the 2010 survey. The new survey compliments and strengthens the 2010 survey in the following ways: (1) 556 respondents were interviewed (more than three times in size compared to that of the 2010 survey) - excluding incomplete responses, outliers and other non–conforming criteria, 453 valid responses used for analysis, compared to 130 in the earlier study; (2) about one-fifth of the respondents were men–owned microenterprises in the new survey compared to none in the previous; (3) the new survey was more heterogeneous compared to the 2010 as it covered five microcredit providers compared to only one previously; and (4) new questions have been added.

There are two main objectives of this paper. First, we are interested to see if the main findings of Atmadja et al. (2016) would prevail with the updated and extended survey data. Second, the updated and extended survey enables us to explore a few more interesting research questions, including, would different lending schemes (group vs. individual) matter for business performance? What about gender? And, does a micro-entrepreneur's ability and/or willingness to separate their personal/consumption expenses from business—related (money separation, hereafter) make any difference to business performance?

The findings of the updated and extended survey are intriguing and yet conforming. Overall, we find that even with a much larger sample, including both men and women—owned microenterprises and greater heterogeneity, microfinance by itself may not necessarily improve business performance. Thus, our findings, by and large, confirm the findings of the Atmadja et al. (2016) study and conform to the strand of literature which argues that microfinance may not really matter for microenterprise performance. With respect to the new research questions, while microcredit schemes (individual versus group) and gender effects may not matter significantly, money separation does appear to have a positive and significant relationship with performance. On the policy front, this paper recommends that non-financial factors such as human capital, spousal involvement, and money separation be considered as important factors for improving microenterprise business performance in the case at least of Indonesia, with less focus on microfinance *per se*.

The rest of this paper is outlined as follows. Section 2 provides a brief literature review—for details, we invite the reader to peruse the relevant sections of the Atmadja et al. (2016) study. Section 3 describes the research methodology. Section 4 provides empirical results with discussion. Section 5 concludes.

2. Brief literature review and hypothesis development

In this section, we develop our hypothesis with respect to each independent variable based on extant literature. With regard to those included in the Atmadja et al. (2016) study, we summarise the literature review and restate the hypotheses since we concur with them—for details, we invite the reader to peruse that study, which can be found in Atmadja et al. (2016). With regard to the three new independent variables—microcredit scheme, gender and money separation—we develop our hypothesis based on a brief review of the literature. Essentially, H_1 to H_3 below are those adopted from the Atmadja et al. (2016) study and H_4 to H_6 are developed in this study.

Hypothesis 1: The relationship between financial capital and microenterprise business performance will be negative.

Hypothesis 2: Human capital will be positively associated with microenterprise business performance.

Hypothesis 3: Social capital will be positively associated with microenterprise business performance.

2.1. Microcredit *schemes* and microenterprise performance

In the microfinance industry, the individual and group lending credit schemes are the most common types available to microenterprises. Under the former, the size of the loan is determined primarily on the basis of the pledged collateral, which may be repossessed in the event of default. Thus, while on a much lower scale, some parallels can be drawn between this and loans obtained from the formal sector such as banks.

Under the latter, microcredit is offered to individuals only via "groups"—an individual borrower applies for a loan through their "lending group". The lending group, assisted by an officer appointed by the microfinance provider, decides on the amount to be approved and subsequently becomes liable for repayment in the event of default². To ensure timely repayment of the loan, the group lending scheme involves frequent repayment meetings and peer pressure. For some borrowers, these might be burdensome, especially if the group is formed from communities with a high degree of social ties. Nevertheless, evidence suggests that the frequent meetings might also benefit the members. For example, using field experiments in India, Feigenberg et al. (2010) provide evidence on the role of microfinance in building social capital by reporting that more frequent group meetings can in practice lead to greater social interactions. These interactions might provide members with alternative sources of information that might not otherwise be available to individuals, which might help discover or create new opportunities in the market (Shepherd et al. 2007). The lending group might develop new or deepen already existing social relationships within the group that might yield economic gains and increase the likelihood of loan repayment (Anthony 2005).

Many microfinance providers in developing economies rely on joint liability for their business operations as a means to induce peer monitoring and reduce ex ante (Stiglitz 1990; Armendariz de Aghion and Morduch 2005; Banerjee et al. 1994) as well as ex post (Besley and Coate 1995; Bhole and Ogden 2010) moral hazard over investment choice particularly in the absence of collateral and the providers' credit screening. However, if trust between members in a lending group is low, along with little enforcement of contracts, this may also become liabilities for the microfinance providers.

The joint liability mechanism of lending groups might also encourage risk taking behaviour by individuals (Giné et al. 2010; G. Fischer 2013). However, the strict peer monitoring practices might, on the other hand, serve to discourage such behaviour (G. Fischer 2013).

The most recent study regarding the impact of microcredit schemes on borrowers has been reported by Attanasio et al. in Tilburg University Discussion Paper No. 2013-074 in 2013. Using a randomized field experiment of 1,148 women who were members of a microfinance institution (MFI) across rural Mongolia, they compared the impact of individual versus group lending microcredit on borrowers. The study finds that microcredit significantly influences business creation in the case of women receiving loans via group lending schemes. The authors argue that one possibility might be that the group lending scheme allows women to mutually insure each other against adverse business outcomes, which may in turn reduce investment uncertainty, increase loan take up, and eventually lead to larger long-run effects such as increase in business performance.

In light of the foregoing,

Hypothesis 4: Group-lending scheme funded microenterprises are likely to perform better than those funded otherwise.

2.2. *Gender* and microenterprise performance

Studies comparing the performance of men and women-owned firms show that those headed by women tend to be generally smaller across various dimensions, including gross revenue, number of employees, sales, assets, and profit levels (E. M. Fischer et al. 1993; Kalleberg and Leicht 1991; Rosa et al. 1996; Watson and Robinson 2003; Ellis et al. 2010). Women-owned enterprises tend to also expand more slowly than those owned by men (Cooper et al. 1994; GEM 2010), which might be due to women being generally more risk-averse (Cliff 1998) and lack relevant industry-specific experience (Loscocco et al. 1991).

Other studies report that women have narrower and more homogenous social networks³ (Loscocco et al. 2009; GEM 2010) and have more limited access to finance (Parker 2009) compared to men. Yet another study, Kessy (2009), finds that men-headed enterprises have higher business performance compared to those headed by women. The study also notes that men in general have higher assets, sales revenue and number of employees than women.

Although some other studies report that women had particularly lower rates of loan default compared to men (Khandker 1998; Khan 1999; Panjaitan-Drioadisuryo and Cloud 1999; Remenyi 2000) which might reduce financial risk to the lenders (Armendariz de Aghion and Morduch 2005), it does not always necessarily mean that women—owned enterprises performed better than those owned by men. The low default rates could instead be due their more risk—averse nature (Velasco and Marconi 2004; Cliff 1998).

Women and men might become entrepreneurs for different reasons as well (Ellis et al. 2010).

While men appear to be more often driven by economic opportunities ("opportunity-motivated"), women

³ Women's social networks emphasize interpersonal relationships over instrumental relationships; hence women are exposed to fewer business-relevant sources (Brush 1992).

tend to be more "necessity-motivated" entrepreneurs who desire to meet the basic necessities or to just find employment. An enterprise helps women improve their earnings capabilities and at the same time enables the desired flexibility of a balanced work—family commitment (Bird and Brush 2002; Brush 1992; Ellis et al. 2010). Consequently, female borrowers tend to have a greater social impact compared to male borrowers because they invest more capital in family education, nutrition and health care (Armendariz de Aghion and Morduch 2005; Blumberg 1988; Panjaitan-Drioadisuryo and Cloud 1999; Pitt et al. 2006).

In light of the foregoing,

Hypothesis 5: Men-owned microenterprises are likely to perform better than women-owned ones.

2.3. Money separation and microenterprise performance

Literature suggests that microcredit often fails to help microenterprises grow because borrowers tend to have limited skills to use borrowed funds effectively (Adams and Von Pischke 1992; Imai et al. 2010; Johnston and Morduch 2008). Evidence shows that borrowers may not use the funds for the intended purpose such as purchasing productive assets, or working capital (Collins et al. 2009; De Mel et al. 2008; Rutherford 2006; Barnes et al. 2001). A possible reason for this inability to distinguish between intended and actual purpose of the loan—or, *money separation* as we refer to in this study—is the lack of access to credit services for the financially disadvantaged to smooth their daily consumption and to deal with emergency as this group are historically thought of as having no need for credit services; hence, microcredit becomes the only credit available to such families (Collins et al. 2009; Hoque 2004).

To provide an example from Indonesia, via a survey of 1,438 households across six provinces of Indonesia in 2002, Johnston and Morduch (2008) find that at least 50% of the microcredit from MFIs to microenterprises were for purposes totally unrelated to business. Instead, the loans were used for home improvement, non–business land or building purchase, school tuition, medical treatment, other loan repayment, daily needs or retirement needs, vehicle purchase, household goods, ceremony or social expenditure, holiday needs, jewellery purchase, etc. Moreover, the authors report that female-headed households were more likely to borrow for daily consumption. The borrowers' inability to distinguish their personal/consumption expenses from business–related investment often prolongs their dependence on external finance (Parker 2009). It is apparent that a good business or management practice, even in the simplest form such as separating money for business from household expenses, might be important for microbusiness success.

In light of the foregoing,

Hypothesis 6: Money separation is positively related to microenterprise business performance.

3. Research method

3.1. The survey

As indicated earlier, the present (2014) survey was conducted in the same location as the previous (2010)—in Indonesia's second largest city of Surabaya and its surroundings, from February to June 2014. Perhaps the first key difference to note between the 2014 and the 2010 surveys is in relation

to the number and extent of micro finance providers covered—the 2010 survey covered only one microfinance provider Setya Bhakti Wanita (SBW); the 2014 survey covered five—four more than the previous, including two Islamic-style microfinance providers⁴ (Madani and Artha Bina Ummat), one cooperative (Assakinah) and one government-sponsored microcredit provider (BKM Bendul Merisi) plus SBW. Table 1 provides a summary of the features of the five providers.

Table 1: Characteristics of the microcredit providers covered in the 2014 survey

	Setya Bhakti Wanita	Artha Bina Ummat	Assakinah	Madani	BKM Bendul Merisi
Established	1978	1998	1999	2007	2010
Туре	Cooperative	Islamic	Cooperative	Islamic	Community Empowerment
Total membership	12470 (all women)	1592 (842 men, 750 women)	971 (51 men, 920 women)	3515 (1439 men 2076 women)	205 (113 men 92 women)
Number of lending groups	418	0	64	0	29
Credit schemes	Individual, Group lending	Individual only	Individual, Group lending	Individual only	Group lending only
Total credit outstanding (in IDR, 2013)	More than 1 billion	More than 1 billion	More than 1 billion	More than 1 billion	Less than 1 billion
Credit ranges (in IDR)	Up to 2b (individual) 1m– 27m (group)	0.5m-40m	Up to 200m (individual) 0.5m to 8m (group)	0.1m to 500m	0.5m to 4.5m
Credit terms	Up to 15 years (individual) 3 – 25 months (group)	Up to 24 months	Up to 24 months (group). 3-6 months (individual)	Up to 48 months	Up to 12 months
Credit requirements	Collateral (individual) Compulsory saving and voluntary saving (group)	Collateral	Collateral (individual) Compulsory saving and voluntary saving (group)	Collateral	Voluntary saving

Source: the data was obtained from the 2014 survey interviews with the MFIs.

As Table 1 shows, our sample provides a reasonable mix of microcredit providers, which in turn provides a good robustness test. For example, the sample includes small, medium and large sized providers, with membership ranging from 205 to 12,470. Similarly, the sample microcredit providers are relatively new (2010) to relatively well–established (1978), cover different types—Islamic, cooperatives and others—and with different combinations of lending group versus individual credit schemes and different make up in terms of men and women memberships.

At the time of the survey, the five microcredit providers had a total membership of 17,553, of which 5,531 (205 from BKM, 3164 from SBW, 738 from Assakinah, 575 from ABU, and 849 from Madani) met the critical survey criteria of "owns a microenterprise". Of the 5,531 who meet the criteria, 1,424 (or 26%) had borrowed via individual credit schemes and the rest, 74%, had borrowed using lending group schemes.

⁴ Since the microcredit providers are formally registered as multipurpose cooperatives and some of their funding sources come from commercial banks' debts which apply standard conventional banking procedures, they might not be able to strictly implement sharia principles on their lending operations.

Of the eligible respondents, those using the lending group schemes to borrow belonged to around 178 lending groups (41 from Assakinah, 108 from SBW, and 29 from BKM). From each of these groups, two to three members were randomly selected as prospective respondents—a total of 530. Respondents using the individual credit scheme were also included in the survey—around 270 were randomly selected as prospective respondents.

Thus, a total of 800 prospective respondents were identified and contacted by the sample microfinance providers on behalf of the researchers for their voluntary participation. Of these, only 556—405 belonging to group lending schemes and 151 individuals—agreed to be interviewed. Interviews were conducted mostly at the respondent's residence or business place to observe their real-life conditions; occasionally the interviews were conducted at group meetings. Of that figure, only 453 completed responses were found to be valid for the purposes of the analysis. Of the 453 respondents, 89 (20%) turned out to be men and 364 (80%) women, a ratio very much reflective of the make—up of the microfinance membership in Indonesia—there are many more women than men.

The interviewers were local university students, from a final year research methods class who voluntarily wanted to participate in the survey. Ten students were selected based on their academic qualifications and relevant prior experience. The researchers provided training to the interviewers prior to the survey, which included technical understanding about the details of the questionnaire and the implementation of the ethical conduct, and closely supervised/monitored them during the data collection process.

3.2. The variables

Dependent variable. Consistent with Atmadja et al. (2016) and for the reasons explained in that study, *profit change* is used as the proxy for measuring business performance.

Independent variables. Again, consistent with Atmadja et al. (2016) and for the reasons explained in that study, we include the following independent variables in our study: microcredit, human capital, and social capital.

In addition, as noted earlier, we include gender (1 = female; 0 otherwise), credit schemes (1 = group lending; 0 otherwise) and money separation (1 = strict separation; 0 otherwise). The term *money separation* in this study is different from the usage in Karlan and Valdivia (2010)⁵. Data was obtained by asking respondents "how do you separate your money for business and money for household expenses?" It was a close-ended question with three optional responses—no separation; not strictly; strictly. The responses were then transformed into a dummy variable (i.e. 1 for strictly, and 0 otherwise).

Control variables. The control variables are change in total assets, competition, number of employees, new products, respondents' age⁶ and length of microfinance membership.

⁵ Karlan and Valdivia (2010) measure money separation as a binary variable equal to 1 if the MFI's client thinks that is not necessary to separate her money from that of her husband or partner or other adult in the household to control expenses and savings. It is treated as a dependent variable.

⁶ For the operational definition of the selected variables, please refer to Atmadja et al. (2016).

3.3. Data description

Table 2 describes the data of selected variables, comparing as well the 2014 survey to the 2010. In the table, data from the 2014 survey are presented in two parts – Part A and Part B.

Table 2. The descriptive statistics of the data.

Varial		2010			20)14		
				Part A			Part B	
			female	male	all	female	male	all
Profit	Decrease	18%	18%	10%	17%	16%	10%	15%
	About the same	34%	19%	8%	18%	20%	9%	18%
	Increase	47%	63%	82%	65%	63%	81%	67%
Microcredit	Mean	11.35	9.98	2.03	8.80	9.66	4.36	8.62
(in millions IDR)								
Education	Elementary	5%	6%	22%	8%	9%	22%	11%
	Junior high	12%	15%	24%	16%	14%	19%	15%
	Senior high	44%	54%	50%	54%	53%	49%	53%
	University	38%	25%	4%	22%	24%	9%	21%
Prior work experience	No	58%	83%	50%	78%	80%	58%	76%
•	Yes	42%	17%	50%	22%	20%	42%	24%
Family business	No	47%	76%	72%	76%	73%	65%	72%
background	Yes	53%	24%	28%	24%	27%	35%	28%
Strongties	Mean	1.99	1.63	1.24	1.57	1.61	1.55	1.60
Weak ties	Mean	7.74	1.68	1.14	1.19	1.16	1.20	1.17
Lending group ties	Mean	31.77	25.42	5.86	22.53	25.42	5.86	22.53
Spousal involvement	Mean	5.18	5.24	4.86	5.18	5.28	5.16	5.25
Change in assets (%)	Mean	5.46	4.95	18.50	6.96	6.12	14.69	7.80
Competition	No	37%	28%	44%	31%	30%	35%	31%
•	Yes	63%	72%	56%	69%	70%	65%	69%
Number of employees	Mean	2.87	0.91	0.82	0.89	0.87	1.10	0.91
New product	Mean	1.97	3.19	2.72	3.12	3.18	3.09	3.17
Age	Mean	48.02	46.53	46.70	46.56	45.41	44.07	45.15
Gender	Male	0%	0%	100%	15%	0%	100%	20%
	Female	100%	100%	0%	85%	100%	0%	80%
Credit scheme	Individual	0%	0%	0%	0%	21%	44%	25%
	Group lending	100%	100%	100%	100%	79%	56%	75%
Length of membership	Mean	10.25	10.77	4.48	9.78	9.43	4.35	8.43
Number of obs.		130	288	50	338	364	89	453

Source: The data are summarised from the 2014 survey conducted by the researchers.

Part A is calculated based on the 2014 data of three microfinance providers offering group lending credit schemes (i.e. Assakinah, SBW, and BKM).

Part B is calculated based on the 2014 data of both credit schemes (i.e. group lending and individual) from the five microcredit providers.

Part A contains the data of three of the five sample microcredit providers offering group lending credit schemes (i.e. Assakinah, BKM, and SBW) with 338 number of observations. Data from this part are used in the analysis meant to address the first study objective, i.e. to compare our findings with that of Atmadja et al. (2016). Part B covers the data of both credit schemes (i.e. group lending and individual) from all of the five sample providers with 453 observations used for the analysis aimed at the second study objective, i.e. to test the influence of schemes (individual vs. group), gender and money separation on microenterprise performance.

In terms of the dependent variables, the table shows that more respondents overall in the 2014 survey experienced growth in profit and sales compared to the 2010 survey. Further, the 2014 respondents generally borrowed less than the 2010's, and men on average borrowed much less compared to women. With respect to human capital (education level, prior working experience and parents/family business background), respondents in the 2010 survey appear to be better overall. Moreover, male respondents were less educated but had more family business background and prior working experience than the female respondents in the both surveys.

Although it is relatively narrower compared to the 2010 respondents, the 2014 female respondents appear to have wider social ties/networks than the male respondents. Compared to findings from other country studies (Loscocco et al. 2009; GEM 2010), it seems that Indonesia female microentrepreneurs tend to not only involve more family members, but also more acquaintances into their networks compared to their male counterparts. The size of women's lending groups is also greater than that of men's.

A concern is also put on the mean of the weakties as it is apparent the 2010 sample has much larger mean than the 2014's. It was suspected that this might be driven by some outliers. However, after re-estimating the model without the seemingly outliers, it was found that the result is not sensitive to those outliers.

4. **Empirical results and discussion**

Since the dependent variables are ordinal (i.e. 1 for decrease, 2 for about the same, and 3 for increase), the ordered probit is appropriately applied for analysis purposes. Table 3 presents the estimation results of the models using the group lending scheme subsample. This is to address the first study objective. In the table, the estimation results of Atmadja et al. (2016) is also restated. The equation used in Atmadja et al. (2016) is applied to estimate the 2014 female sample.

Overall, when the dependent variable is profit change, both estimations yield similar results, indicating that the empirical result is not too sensitive across the two samples. Loan has a significant negative relationship with profit change in 2010 (at 10% level) and 2014 (at 5% level). Spouse involvement is the only social capital variable that matters for profit in 2014 as well as in 2010. The association of the both variables is more significant in 2014 (at 1% level) than in 2010 (at 10% level). Nevertheless, the human capital variables have indifferent results. Education significantly matters in 2010, but does not in 2014. Instead, prior work experience influences are significant to profit in 2014.

Table 3. Ordered probit estimation results of group lending samples

Coef Coef Coef	Table 3. Ordered probit estimation res		•
mancial capital icrocredit -0.0645** -0.0271* (0.0275) (0.0142) man capital lucation 0.2204* 0.0810 (0.1344) (0.0987) ior working experience -0.1043 0.4278* (0.2333) (0.2517) mily business background 0.0755 -0.2680 (0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319	Variables	2010 ¹	2014 ²
icrocredit -0.0645** -0.0271* (0.0275) (0.0142) man capital lucation 0.2204* 0.0810 (0.1344) (0.0987) ior working experience -0.1043 0.4278* (0.2333) (0.2517) mily business background 0.0755 -0.2680 (0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319		Coei	Coei
icrocredit -0.0645** -0.0271* (0.0275) (0.0142) man capital lucation 0.2204* 0.0810 (0.1344) (0.0987) ior working experience -0.1043 0.4278* (0.2333) (0.2517) mily business background 0.0755 -0.2680 (0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0319	Financial capital		
(0.0275) (0.0142) man capital lucation 0.2204* 0.0810 (0.1344) (0.0987) for working experience -0.1043 0.4278* (0.2333) (0.2517) mily business background 0.0755 -0.2680 (0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319	Microcredit	-0.0645**	-0.0271*
10 10 10 10 10 10 10 10		(0.0275)	(0.0142)
10 10 10 10 10 10 10 10			
(0.1344) (0.0987) ior working experience -0.1043 0.4278* (0.2333) (0.2517) mily business background 0.0755 -0.2680 (0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319	Human capital		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Education		
(0.2333) (0.2517) mily business background (0.2755 -0.2680 (0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 (0.0536 (0.0627) eak ties -0.0118 -0.0319		` ′	
mily business background 0.0755 -0.2680 (0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 (0.0627) -0.0118 -0.0319	Prior working experience		
(0.2274) (0.1995) cial capital rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319		` /	
cial capital rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319	Family business background		
rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319		(0.2274)	(0.1995)
rong ties -0.0138 -0.0685 0.0536 (0.0627) eak ties -0.0118 -0.0319	Social capital		
eak ties -0.0118 -0.0319	Strong ties	-0.0138	-0.0685
		0.0536	(0.0627)
0.0108 (0.0496)	Weak ties	-0.0118	-0.0319
(*****)		0.0108	(0.0496)

Lending group (LG-ties)	0.0171	0.0052
	(0.0146)	(0.0089)
Spousal involvement	0.1332*	0.1292***
	(0.0719)	(0.0500)
Business Control		
Change in assets	0.0254**	0.0408***
	(0.0062)	(0.0077)
Competition	0.4968**	-0.5134**
	(0.2369)	(0.2010)
Number of employees	0.0391	-0.0396
	(0.0506)	(0.0586)
New product	0.0706	0.0241
	(0.1197)	(0.0479)
Individual control		
Age	0.0011	0.0131
	(0.0139)	(0.0133)
Length of membership	0.0294	-0.0068
	(0.0189)	(0.0131)
Number of observation	130	288

Note: Dependent variable is profit change (1 if decrease, 2 if about the same, and 3 if increase) used as a proxy of business performance. Unstandardized coefficients and standard error (in parentheses) are presented.

To address the second study objective, five models are estimated using the 2014 data of the both credit schemes. In these models, while business performance remains the dependent variable, lending group ties variable is omitted from the analysis and all covariates used in Table 3 are treated as control variables. In addition, institution dummies (i.e. Arta Bina Umat, Madani, Assakinah, and BKM) and types of industry (i.e. manufacture, wholesale/retail, and services) are included as control variables in the models. The estimation results of the models are presented in Table 4.

In Model 1, microenterprise business performance is regressed on all control variables to provide a baseline model. Table 4 shows that microcredit and the other four variables (i.e. prior work experience, spousal involvement, assets change, and competition) are significantly linked to business performance. Corresponding to $H_4 - H_6$, credit scheme is then included as a covariate in Model 2, gender in Model 3, and money separation in Model 4. Finally, business performance is regressed on all variables in Model 5.

Table 4. Ordered probit estimation results of 2014 samples.

Variables	Model 1	Model 2	Model 3	Model 4	Model 5
Control variables					
Microcredit	-0.0248**	-0.0248**	-0.0244**	-0.0257**	-0.0254**
	(0.0117)	(0.0117)	(0.0117)	(0.0118)	(0.0118)
Education	0.0659	0.0673	0.0690	0.0582	0.0639
	(0.0768)	(0.0770)	(0.0771)	(0.0771)	(0.0776)
Prior working experience	0.5140***	0.5120***	0.5240***	0.5540***	0.5590***
	(0.1830)	(0.1830)	(0.1830)	(0.1830)	(0.1830)
Family business background	-0.1950	-0.1960	-0.1930	-0.2010	-0.2010

^{***} p<0.01, ** p<0.05, * p<0.1.

¹ Restated from Atmadja et al. (2016).

² Estimated using female samples of the 2014 data from three microfinance providers offering group lending credit schemes (i.e. Assakinah, SBW, and BKM).

	(0.1570)	(0.1570)	(0.1570)	(0.1570)	(0.1570)
Strong ties	-0.0399	-0.0402	-0.0451	-0.0100	-0.0151
	(0.0563)	(0.0563)	(0.0564)	(0.0571)	(0.0573)
Weak ties	0.0032	0.0037	-0.0007	0.0020	-0.0013
	(0.0438)	(0.0438)	(0.0438)	(0.0438)	(0.0439)
Spousal involvement	0.1320***	0.1320***	0.1330***	0.1260***	0.1270***
	(0.0421)	(0.0421)	(0.0422)	(0.0424)	(0.0425)
Change in assets	0.0317***	0.0319***	0.0321***	0.0298***	0.0304***
	(0.0049)	(0.0049)	(0.0049)	(0.0049)	(0.0051)
Competition	-0.4480***	-0.4520***	-0.4560***	-0.4410***	-0.4560***
	(0.1580)	(0.1590)	(0.1580)	(0.1600)	(0.1610)
Number of employees	-0.0206	-0.0210	-0.0326	-0.0328	-0.0455
	(0.0485)	(0.0485)	(0.0488)	(0.0489)	(0.0493)
New product	0.0428	0.0435	0.0375	0.0279	0.0230
	(0.0394)	(0.0395)	(0.0395)	(0.0401)	(0.0404)
Age	0.0119	0.0120	0.0114	0.0119	0.0114
	(0.0100)	(0.0100)	(0.0100)	(0.0101)	(0.0101)
Length of membership	-0.0136	-0.0142	-0.0136	-0.0086	-0.0095
	(0.0117)	(0.0119)	(0.0117)	(0.0119)	(0.0122)
Arta Bina Umat	-0.1790	-0.0933	-0.3370	-0.0791	-0.0841
	(0.2260)	(0.3940)	(0.2460)	(0.2290)	(0.4120)
Madani	0.3370	0.4210	0.1430	0.4060	0.3710
	(0.2850)	(0.4280)	(0.3060)	(0.2910)	(0.4490)
Assakinah	-0.2560	-0.2670	-0.2520	-0.1990	-0.2120
	(0.2110)	(0.2140)	(0.2110)	(0.2130)	(0.2170)
BKM	-0.1980	-0.2100	-0.6830	-0.0147	-0.5200
	(0.2880)	(0.2920)	(0.4180)	(0.2940)	(0.4260)
Manufacture	0.0250	0.0268	0.0314	0.0222	0.0318
	(0.1710)	(0.1710)	(0.1710)	(0.1730)	(0.1730)
Trading	0.0201	0.0242	0.0315	-0.0208	-0.0018
<u> </u>	(0.1670)	(0.1670)	(0.1670)	(0.1690)	(0.1700)
	,	`(').		,	,
Main covariates					
Lending scheme (LG=1)		0.0955			0.1710
,		(0.360)			(0.3650)
Gender (female=1)		` /	-0.4780		-0.4800
,			(0.2980)		(0.3000)
Money separation				0.4970***	0.5020***
~ x				(0.1550)	(0.1550)
				, ,	` '
Number of observation	453	453	453	453	453

Note: Dependent variable is profit change (1 if decrease, 2 if about the same, and 3 if increase) used as a proxy of business performance. Unstandardized coefficients and standard error (in parentheses) are presented.

*** p<0.01, ** p<0.05, * p<0.1.

The models are estimated using the 2014 data of the both credit schemes from the five microcredit providers.

Table 4 generally shows that, in all models, microcredit has negative effect on profit, while the other variables have positive impact on the business performance partially confirming hypothesis 2 and hypothesis 3. These findings are to some extent comparable with those of Atmadja et al. (2016) suggesting that the results are not sensitive to sample size and the inclusion of the three new variables (i.e. credit scheme, gender and money separation).

In Model 2 and Model 5 of the tables, while the credit scheme (group vs. individual) shows a positive relationship with business performance, the association is statistically insignificant, suggesting that the type of scheme might not really matter for a microenterprise's business performance; hence rejecting hypothesis 4. This suggests that social capital created through a lending group membership may

not necessarily make a significant difference to business performance. It is possibly because the agenda of group meetings have not been able to adequately capture business related matters—have largely been dominated by loan repayment issues so thus neglected the meaningful business related discussions.

Secondly, in some microfinance providers, the group lending size was too large (up to 50 members per group). On the one hand, this can broaden business/social networks among the members. On the other hand, this could adversely affect the quality of peer credit screening and monitoring within groups. Particularly in the situation when numerous loan applications needed to be reviewed promptly, the group's considerations were more often based on credit ceiling and track records of the applicants with little attention paid to the purpose of the loan. This might stimulate risk taking behaviour of the applicants (Giné et al. 2010), which possibly ends up with either business success or business failure. It might be true that group lending has a positive impact on business creation; however it does not necessarily mean that the performance will always be better.

This study also finds that there is no significant gender effect on microenterprises business performance, as shown in Model 3 and Model 5. Although the negative sign of the variable's coefficient might indicate that men-owned microenterprises are more likely to perform better than women-owned ones, the difference is not significant statistically. This rejects hypothesis 5.

While credit schemes and gender do not significantly matter for performance, money separation apparently has an important role. The result of Model 4 and Model 5 in Table 4 show that the variable's coefficients are significant at 1% level with positive signs confirming hypothesis 6. Thus, holding other variables unchanged, microenterprises owned by micro-entrepreneurs who strictly separate their money for business from household use are more likely to experience better business performance compared to others. This finding provides an empirical support to the prior studies stating that borrowers' lack of self-discipline in using the loan could results in poor business performance and long-term dependence on external funding (Parker 2009; Adams and Von Pischke 1992; Imai et al. 2010; Johnston and Morduch 2008).

With relatively higher interest rate imposed on the loan, the inability of micro-entrepreneurs to properly manage the loan might cause a financial burden to the enterprises, especially if they are not able to generate higher profits to compensate the relatively higher interest costs, leading to long-term dependence on external finance. Conversely, if the entrepreneurs are able to strictly separate the money, they might have sufficient reserves for financing working capital and/or purchasing business assets to expand their business to generate more profit. This basic business practice appears to have benefited the performance of Indonesian microenterprises.

Considering the above findings, an implication that can be drawn is that microcredit *per se* might not enhance microenterprise business performance. The lack of self-discipline or inability of micro entrepreneurs to adequately manage finances might be a key factor, highlighting the importance of business and financial management skills for microenterprise success. To enhance business performance, micro entrepreneurs should not only be provided with easy access to finance but also be equipped with sufficient abilities to appropriately manage their business. For example, evidence from Tanzania shows that enterprises whose owners received business training and extension advice performed better than

those that did not (Kuzilwa 2005). Accordingly, this study suggests that training in business and management should ideally become the complements of microcredit. The other non-financial factors (i.e. human capital and spousal involvement) might also be considered in designing credit policies particularly to make sure that the credit can be delivered effectively with minimum risks.

5. Conclusion

Essentially, this study attempts to mitigate the limitations of a recent study by Atmadja et al. (2016), which apparently was the first to test the effect of microcredit on the performance of microenterprises in the case of Indonesia and finds a negative relationship. The limitations of that study, based on a survey of women microenterprises undertaken in 2010 in Surabaya, Indonesia's second largest city, included relatively small sample, lacked heterogeneity, and focussed only on women. The present study is based on an extended and updated survey of microenterprises in Surabaya—with a larger, more heterogeneity sample, and includes men owned enterprises.

In addition to testing the main findings of the Atmadja et al. (2016) study, we investigate if different lending schemes (group vs. individual), gender and a micro-entrepreneurs' ability and/or willingness to separate their personal/consumption expenses from business—which we call "money separation"—might matter for business performance. Overall, we find that even with a much larger sample, including both men and women—owned microenterprises and greater heterogeneity, microfinance by itself may not necessarily improve business performance. Thus, our findings, by and large, confirm the findings of the Atmadja et al. (2016) study and conform to the strand of research which argues that microfinance may not really matter for microenterprise business performance. We also find that while microcredit schemes (individual versus group) and gender effects may not matter significantly, money separation does appear to have a positive and significant relationship with performance. Thus, on the policy front, this paper recommends that non-financial factors such as human capital, spousal involvement, and money separation be considered as important factors for improving microenterprise business performance in the case at least of Indonesia, with less focus on microfinance *per se*.

Future studies might involve an even larger, more balance and more heterogeneous sample size from across Indonesia. The inclusion of more male respondents and more variables affecting the microenterprises' business performance might be taken into consideration as well. It might also be necessary to provide a deeper explanation of the relationships noted in this study, for example, why loan size has a negative effect on profit but not on sales, why social ties/networks do not matter to performance, and whether applying alternative credit schemes (e.g. individual credit scheme) might differently affect business performance.

References

Adams, D. W., & Von Pischke, J. D. (1992). Microenterprise Credit Programs: Déja vu. World Development, 20(10), 1463-1470.

Anthony, D. (2005). Cooperation in Microcredit Borrowing Groups: Identity, Sanctions, and Reciprocity in the Production of Collective Goods. *American Sociological Review, 70*(3), 496-515.

- Armendariz de Aghion, B., & Morduch, J. (2005). *The Economics of Microfinance*. Cambridge, MA: The MIT Press.
- Atmadja, A.S., Su, J.J., & Sharma, P. (2016). Examining the Impact of Microfinance on Microenterprise Performance (Implication for Women-owned Microenterprises in Indonesia). *International Journal of SocialEconomics*, 43(10), 962-981.
- Banerjee, A. V., Besley, T., & Guinnane, T. W. (1994). The Neighbor's Keeper: The Design of a Credit Cooperative with Theory and a Test. *The Quarterly Journal of Economics*, 109(2), 491-515.
- Barnes, C., Keogh, E., & Nemarundwe, N. (2001). Microfinance Program Clients and Impact: An Assessment of Zambuko Trust, Zimbabwe. Washington, DC: AIMS.
- Bateman, M., & Chang, H. J. (2012). Microfinance and the illusion of development: from hubris to nemesis in thirty years. *World Economic Review, 1*, 13-36.
- Besley, T., & Coate, S. (1995). Group Lending, Repayment Incentives and Social Collateral. *Journal of Development Economics*, 46(1), 1-18.
- Bhole, B., & Ogden, S. (2010). Group Lending and Individual Lending with Strategic Default. *Journal of Development Economics*, 91(2), 348-363.
- Bird, B., & Brush, C. G. (2002). A Gendered Perspective on Organizational Creation. [Article]. Entrepreneurship Theory and Practice, 26(3), 41-65.
- Blumberg, R. L. (1988). Income Under Female Versus Male Control: Hypotheses from a Theory of Gender Stratification and Data from the Third World. *Journal of Family Issues*, 9(1), 51-84.
- Bradley, S. W., McMullen, J. S., Atmadja, A. S., Simiyu, E. M., & Artz, K. (2011). Self-employed or Employed Others? Pre-Entry Capabilities, Entreprenerial Action, and the Learned Resourcefulness of Microcredit Firm Founders. Frontiers of Entrepreneurship Research, 31(19), 644-658.
- Brush, C. G. (1992). Research on Women Business Owners: Past Trends, a New Perspective and Future Directions. *Entrepreneurship: Theory & Practice*, 16(4), 5-30.
- Badan Kredit Desa (BKDs) (2013). Banking With The Poor Network. http://www.bwtp.org/arcm/indonesia/II_Organisations/MF_Providers/BKDs.htm. Accessed 21/02/2013.
- Chen, M. A., & Snodgrass, D. (2001). Managing Resources, Activities, and Risk in Urban India: The Impact of SEWA Bank. Washington, DC: AIMS.
- Cliff, J. E. (1998). Does One Size Fit All? Exploring the Relationship between Attitudes towards Growth, Gender, and Business Size. *Journal of Business Venturing*, 13(6), 523-542.
- Collins, D., Morduch, J., Rutherford, S., & Ruthven, O. (2009). *Portfolios of the Poor: How the World's Poor Live on \$2 a Day*. Princeton, NJ: Princeton University Press.
- Cooper, A. C., Gimeno-Gascon, F. J., & Woo, C. Y. (1994). Initial Human and Financial Capital as Predictors of New Venture Performance. *Journal of Business Venturing*, 9(5), 371-395.
- Copestake, J., Bhalotra, S., & Johnson, S. (2001). Assessing the Impact of Microcredit: A Zambian Case Study. [Article]. *Journal of Development Studies*, *37*(4), 81-100.
- Crépon, B., Devoto, F., Duflo, E., & Parienté, W. (2011). Impact of Microcredit in Rural Areas of Morocco: Evidence from a Randomized Evaluation. *MIT Working Paper*. Cambridge, MA: MIT.
- Davis, M. (2006). Planet of Slums: Urban Involution and the Informal Working Class. New York: Verso.
- De Mel, S., McKenzie, D., & Woodruff, C. (2008). Returns to Capital in Microenterprises: Evidence from a Field Experiment. *The Quarterly Journal of Economics*, 123(4), 1329-1372.
- Dunn, E., & Arbuckle Jr, J. G. (2001). The Impacts of Microcredit: A Case Study from Peru. *USAID AIMS Project*. Washington DC: USAID Office of Microenterprise Development.
- Ellis, A. N., Orlando, M. B., Muñoz Boudet, A. M., Piras, C., Reimao, M., Cutura, J., et al. (2010). Women's Economic Opportunities in the Formal Private Sector in Latin America and the Caribbean: A Focus on Entrepreneurship. Inter-American Development Bank.
- Feigenberg, B., Field, E. M., & Pande, R. (2010). Building Social Capital through Microfinance. *NBER Working Paper Series*: National Bureau of Economic Research.
- Fischer, E. M., Reuber, A. R., & Dyke, L. S. (1993). A Theoretical Overview and Extension of Research on Sex, Gender, and Entrepreneurship. *Journal of Business Venturing*, 8(2), 151-168.
- Fischer, G. (2013). Contract Structure, Risk-Sharing, and Investment Choice. *Econometrica*, 81(3), 883-939.
- GEM (2010). 2010 Report: Women Entrepreneurs Worldwide: Global Entrepreneurship Monitor.
- Giné, X., Jakiela, P., Karlan, D., & Morduch, J. (2010). Microfinance Games. *American Economic Journal: Applied Economics*, 2(3), 60-95.
- Holloh, D. (2001). ProFi Microfinance Institutions Study. Jakarta: ProFI.

- Hoque, S. (2004). Micro-credit and the Reduction of Poverty in Bangladesh. *Journal of Contemporary Asia*, 34(1), 21-32.
- Imai, K. S., Arun, T., & Annim, S. K. (2010). Microfinance and Household Poverty Reduction: New Evidence from India. *World Development*, 38(12), 1760–1774.
- Johnston, D. J., & Morduch, J. (2008). The Unbanked: Evidence from Indonesia. *The World Bank Economic Review*, 22(3), 517-537.
- Kalleberg, A. L., & Leicht, K. T. (1991). Gender and Organizational Performance: Determinants of Small Business Survival and Success. *The Academy of Management Journal*, 34(1), 136-161.
- Karlan, D., & Valdivia, M. (2010). Teaching Entrepreneurship: Impact of Business Training on Microfinance Clients and Institutions. *Review of Economics and Statistics*, 93(2), 510-527.
- Kessy, S. Microfinance and Enterprises Performance in Tanzania: Does Gender Matter? In A paper presented at 10th Annual IAABD Conference in, 2009: International Academy of African Business and Development (IAABD)
- Khan, M. R. (1999). Microfinance, Wage Employment and Housework: A Gender Analysis. *Development in Practice*, 9(4), 424-436.
- Khandker, S. R. (1998). Fighting Poverty with Microcredit: Experience in Bangladesh. Oxford: Oxford University Press.
- Kondo, T., Orbeta, A., Dingcong, C., & Infantado, C. (2008). Impact of Microfinance on Rural Households in the Philippines. *IDS Bulletin*, 39(1), 51-70.
- Kuzilwa, J. A. (2005). The Role of Credit for Small Business Success: A Study of the National Entrepreneurship Development Fund in Tanzania. *Journal of Entrepreneurship*, 14(2), 131-161.
- Loscocco, K. A., Monnat, S. M., Moore, G., & Lauber, K. B. (2009). Enterprising Women: A Comparison of Women's and Men's Small Business Networks. *Gender & Society*, 23(3), 388-411.
- Loscocco, K. A., Robinson, J., Hall, R. H., & Allen, J. K. (1991). Gender and Small Business Success: An Inquiry into Women's Relative Disadvantage. *Social Forces*, 70(1), 65-65.
- Olson, P. D., Zuiker, V. S., Danes, S. M., Stafford, K., Heck, R. K. Z., & Duncan, K. A. (2003). The Impact of the Family and the Business on Family Business Sustainability. *Journal of Business Venturing*, 18(5), 639-666.
- Panjaitan-Drioadisuryo, R. D. M., & Cloud, K. (1999). Gender, Self-employment and Microcredit Programs: An Indonesian Case Study. *The Quarterly Review of Economics and Finance, 39*(5), 769-779
- Parker, S. C. (2009). The Economics of Entrepreneurship. New York: Cambridge University Press.
- Pitt, M. M., Khandker, S. R., & Cartwright, J. (2006). Empowering Women with Micro Finance: Evidence from Bangladesh. *Economic Development and Cultural Change*, 54(4), 791-831.
- Remenyi, J. (2000). Is There a 'State of the Art' in Microfinance? In J. Remenyi, & B. Quinoñes (Eds.), Microfinance and Poverty Alleviation: Case Studies from Asia and the Pacific (pp. 25-61). London: Pinter.
- Rosa, P., Carter, S., & Hamilton, D. (1996). Gender as a Determinant of Small Business Performance: Insights from a British Study. *Small Business Economics*, 8(6), 463-478.
- Rutherford, S. (2006). Uses and Users of MFI Loans in Bangladesh. *MicroSave. Briefing Notes on Grameen II*: MicroSave.
- Shepherd, D. A., McMullen, J. S., & Jennings, P. D. (2007). The Formation of Opportunity Beliefs: Overcoming Ignorance and Reducing Doubt. *Strategic Entrepreneurship Journal*, 1(1-2), 75-95.
- Stiglitz, J. E. (1990). Peer Monitoring and Credit Markets. *The World Bank Economic Review*, 4(3), 351-366.
- Su, J. J., Atmadja, A., & Sharma, P. (2015). Financial, Human, and Social Capital: What Matters More for Women's Microenterprises? A Case Study of Indonesia. Brisbane: Griffith University. Available at: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2667312.
- Velasco, C., & Marconi, R. (2004). Group Dynamics, Gender and Microfinance in Bolivia. *Journal of International Development*, 16(3), 519-528.
- Watson, J., & Robinson, S. (2003). Adjusting for Risk in Comparing the Performances of Male- and Female-controlled SMEs. *Journal of Business Venturing*, 18(6), 773-788.
- Wood, M. S., Bradley, S. W., & Artz, K. (2015). Roots, Reasons, and Resources: Situated Optimism and Firm Growth in Subsistence Economies. *Journal of Business Research*, 68(1), 127-136.

Bukti korespondensi permintaan revisi oleh reviewer (14 September 2017)

Fwd: International Journal of Social Economics - Decision on Manuscript ID IJSE-02-2017-0031

Parmendra Sharma <p.sharma@griffith.edu.au>

Jum 15/09/2017 06.21

Kepada: adwin.atmadja <adwin.atmadja@griffithuni.edu.au>

Cc: JJ Su <j.su@griffith.edu.au>

1 2 lampiran (470 KB)

paper 1b.docx; Paper 1b-IJSE-1.2.17 (1).pdf;

Hi Adwin,

Please see email below--this is good news--let's revise and re-submit. Can you please have a first go at the revision? Please copy the comments into a new file/doc, in a table with columns for reviewer's comments, our response and page number. In the revised manuscript, please use track changes for now.

The revisions are due on 14 Nov--can you do this by end of the month (30 Sept) or earlier? I've attached two versions of the paper: (i) submitted PDF; and (ii) the only Word copy I have--would you have a more updated version?

Before you start the revisions, let's make sure we are using the right Word version.

Cheers, PS

----- Forwarded message -----

From: International Journal of Social Economics < onbehalfof+J.Connelly+hull.ac.uk@manuscriptcentral.com>

Date: 14 September 2017 at 22:32

Subject: International Journal of Social Economics - Decision on Manuscript ID IJSE-02-2017-0031

To: p.sharma@griffith.edu.au

14-Sep-2017

Dear Dr. Sharma:

Manuscript ID IJSE-<u>02-2017-0031</u> entitled "Microfinance and microenterprise performance in Indonesia: an extended and updated survey" which you submitted to the International Journal of Social Economics, has been reviewed. The comments of the reviewer(s) are included at the bottom of this letter.

The reviewer(s) have recommended publication, but also suggest some revisions to your manuscript. Therefore, I invite you to respond to the reviewer(s)' comments and revise your manuscript.

To revise your manuscript, log into https://mc.manuscriptcentral.com/ijsec and enter your Author Centre, where you will find your manuscript title listed under "Manuscripts with Decisions." Under "Actions," click on "Create a Revision." Your manuscript number has been appended to denote a revision.

You will be unable to make your revisions on the originally submitted version of the manuscript. Instead, revise your manuscript using a word processing program and save it on your computer. Please also highlight the changes to your manuscript within the document by using the track changes mode in MS Word or by using bold or coloured text. Once the revised manuscript is prepared, you can upload it and submit it through your Author Centre.

When submitting your revised manuscript, you will be able to respond to the comments made by the reviewer(s) in the space provided. You can use this space to document any changes you make to the original manuscript. In order to expedite the processing of the revised manuscript, please be as specific as possible in your response to the reviewer(s).

Emerald has partnered with Peerwith to provide authors with expert editorial support, including language editing and translation, visuals, and consulting. If your article was rejected, or had major revisions requested on the basis of the language or clarity of communication, you might benefit from a Peerwith expert's input. For a full list of Peerwith services, visit: https://authorservices.emeraldpublishing.com/

Please note that there is no obligation to use Peerwith and using this service does not guarantee publication.

IMPORTANT: Your original files are available to you when you upload your revised manuscript. Please delete any redundant files before completing the submission.

Because we are trying to facilitate timely publication of manuscripts submitted to the International Journal of Social Economics, your revised manuscript should be uploaded as soon as possible. If it is not possible for you to submit your

revision in a reasonable amount of time, we may have to consider your paper as a new submission.

Once again, thank you for submitting your manuscript to the International Journal of Social Economics and I look forward to receiving your revision.

Sincerely,
Prof. James Connelly
Editor, International Journal of Social Economics
J.Connelly@hull.ac.uk

Reviewer(s)' Comments to Author:

Reviewer: 1

Recommendation: Major Revision

Comments: None.

Additional Questions:

1. Originality: Does the paper contain new and significant information adequate to justify publication?: The paper is interesting. This has tried to follow up a study conducted by Atmadja and et al using the survey data of 2010 and their result was published in 2016. This paper does not position well why such a study needs to be repeated although few new dimensions such as group vs individual lending, gender and separation of personal and business related consumption. The paper is written and presented in a clumsy way that restricts a clear decision whether it merits for publications. What is new and how it is different from the findings of other scholars, who have already contributed that microfinance has both positive as well as negative impacts on micro entrepreneurs.

What is the objective of the paper?

-

- 3. Methodology: Is the paper's argument built on an appropriate base of theory, concepts, or other ideas? Has the research or equivalent intellectual work on which the paper is based been well designed? Are the methods employed appropriate?: Since it is a follow up study, clarification is needed how this methodology is improved. Difference between the previous methodology and current methodology should be discussed with advantages and disadvantages. Larger sample size is advantage but the sampling procedure and dividing the sample size logically between different categories such as group vs individual, or a mixture of both (may overlapping in this case), gender, and others are not clear. There are different sources of microfinace with different sets of rules and regulations applicable to group and individuals. Treating samples together either in group or individual category, so also in gender groups can not be justifiable. The major issue raised for the separation of microfinance for personal consumption and business use is not transferred to real variables to address objectively.

There are several hypotheses.

Can it be summed up with one research hypothesis with few sub-hypotheses?

Methodology must follow the objective of the paper and hypothesis in a logical way to reach the conclusion. The paper given an impression that it is compressed from a larger study with many other important facts behind this study. Clear expression of variables (qualitative and quantitative) is remained opaque and gives the chance for speculation and interpretation by individual readers. This must be addressed.

-

- 5. Implications for research, practice and/or society: Does the paper identify clearly any implications for research, practice and/or society? Does the paper bridge the gap between theory and practice? How can the research be used in practice (economic and commercial impact), in teaching, to influence public policy, in research (contributing to the body of knowledge)? What is the impact upon society (influencing public attitudes, affecting quality of life)? Are these implications consistent with the findings and conclusions of the paper?: Not highlighted.
- <bs/>

Bukti konfirmasi submit revisi dan artikel yang direvisi (26 Oktober 2017)

International Journal of Social Economics - Manuscript ID IJSE-02-2017-0031.R1

International Journal of Social Economics <onbehalfof+j.pearce+2010.hull.ac.uk@manuscriptcentral.com>
Kam 26/10/2017 09.55

Kepada: aplin@petra.ac.id <aplin@petra.ac.id>;adwin.atmadja@griffithuni.edu.au <adwin.atmadja@griffithuni.edu.au>;j.su@griffith.edu.

Dear Dr. Sharma:

Your revised manuscript entitled "Microfinance and microenterprise performance in Indonesia: an extended and updated survey" has been successfully submitted online and is presently being given full consideration for publication in the International Journal of Social Economics.

Your manuscript ID is IJSE-02-2017-0031.R1.

Please mention the above manuscript ID in all future correspondence or when calling the office for questions. If there are any changes in your street address or e-mail address, please log in to Manuscript Central at https://mc.manuscriptcentral.com/ijsec and edit your user information as appropriate.

You can also view the status of your manuscript at any time by checking your Author Centre after logging in to https://mc.manuscriptcentral.com/ijsec

Thank you for submitting your manuscript to the International Journal of Social Economics.

Sincerely,

International Journal of Social Economics Editorial Office

P.S. Don't forget, you can read the journal online at http://www.emeraldinsight.com/0306-8293.htm



International Journal of Social Economics

Microfinance and microenterprise performance in Indonesia: an extended and updated survey

Journal:	International Journal of Social Economics
Manuscript ID	IJSE-02-2017-0031.R1
Manuscript Type:	Research Paper
Keywords:	microfinance, microenterprise performance, money separation

SCHOLARONE™ Manuscripts

Microfinance and microenterprise performance in Indonesia: an extended and updated survey

Structured abstract

Purpose

This study addresses the small, women micro-entrepreneur dominated and heterogeneity limitations of the Atmadja et al. (2016) study. The sample is much larger, includes more men and is more heterogeneous, which allows deeper insights and more meaningful explanation of the relationship between microfinance and microenterprise performance in the case of Indonesia, including the effects of gender, lending scheme and money separation.

Design/methodology/approach

Survey of 556 respondents across five micro-credit providers in the city of Surabaya using updated instrument. Ordered probit is used to analyse data.

Findings

Microfinance may not matter for microenterprise performance in the case of Indonesia. Additionally, microcredit schemes (individual versus group) and gender may also not matter for performance, but money separation might have some influence.

Practical implications

Non-financial factors such as human capital, spousal involvement, and money separation should be considered as important factors for improving microenterprise business performance in Indonesia, with less focus on microcredit per se.

Originality/value

This study provides further evidence that microfinance may not matter for microenterprise performance in the case of Indonesia, a populous middle income country with a very long history of microfinance.

Keywords: microfinance, microenterprise, performance, microcredit schemes, gender, money separation.

JEL code: O160

1. Introduction

Debates abound and long-standing on the effects of microfinance on microenterprise performance; see for example Barnes, Keogh, and Nemarundwe (2001), Chen and Snodgrass (2001), Copestake, Bhalotra, and Johnson (2001), Dunn and Arbuckle Jr (2001), Kondo, Orbeta, Dingcong, and Infantado (2008), and Crépon, Devoto, Duflo, and Parienté (2011) who argue for a positive relationship and Adams and Von Pischke (1992), Olson et al. (2003), Davis (2006), and Bateman and Chang (2012) who argue that microfinance may not really matter for performance.

Despite a very long history of microfinance in Indonesia¹—a populous, middle income county bestowed with constant and concerning levels of poverty and economic challenges—little remains known about the effects of microfinance on microenterprise performance in the country. To the best of our

¹ The Bank Priyayi of Purwokerto established in 1895 has been known as the first formal commercial microfinance institution in Indonesia (BWTP, 2013; Holloh, 2001).

knowledge, the most relevant study so far might be by Atmadja et al. (2016) who use data obtained from a 2010 survey to provide an in-depth understanding of the role of microfinance for microenterprise performance in Surabaya, the second largest city in Indonesia. Essentially, the findings suggest that microcredit itself does not necessarily improve business performance.

While the 2010 survey data has also been used in other published works such as Bradley, McMullen, Atmadja, Simiyu, and Artz (2011), Wood, Bradley, and Artz (2015), Atmadja et al. (2016) and Neubert, Bradley, Ardianti, and Simiyu (2017) acknowledge a number of limitations of the survey with respect to their specific investigation, i.e. testing the relationship between microfinance and microenterprise performance, and suggest that a larger sample, including men—owned microenterprises and greater heterogeneity might provide deeper insights and more meaningful explanation of the relationship between microfinance and microenterprise performance.

The above has exactly been the motivation of the present study. Essentially, we have gone back to the city of Surabaya and conducted a much wider and larger survey—we have extended and updated the 2010 survey. The new survey compliments and strengthens the 2010 survey in the following ways: (1) 556 respondents were interviewed (more than three times in size compared to that of the 2010 survey) - excluding incomplete responses, outliers and other non–conforming criteria, 453 valid responses used for analysis, compared to 130 in the earlier study; (2) about one-fifth of the respondents were men–owned microenterprises in the new survey compared to none in the previous; (3) the new survey was more heterogeneous compared to the 2010 as it covered five microcredit providers compared to only one previously; and (4) new questions have been added.

There are two main objectives of this paper. First, we are interested to see if the main findings of Atmadja et al. (2016) would prevail with the updated and extended survey data. Second, the updated and extended survey enables us to explore a few more interesting research questions, which have not been addressed previously in the context of microfinance, including, would different lending schemes (group vs. individual) matter for business performance? What about gender? And, does a micro-entrepreneur's ability and/or willingness to separate their personal/consumption expenses from business—related (money separation, hereafter) make any difference to business performance?

The findings of the updated and extended survey are intriguing and yet conforming. Overall, we find that even with a much larger sample, including both men and women–owned microenterprises and greater heterogeneity, microfinance by itself may not necessarily improve business performance. Thus, our findings, by and large, confirm the findings of the Atmadja et al. (2016) study and conform to the strand of literature (e.g. Adams and Von Pischke (1992), Imai, Arun, and Annim (2010), Johnston and Morduch (2008)) which argues that microfinance may not really matter for microenterprise performance. With respect to the new research questions, while microcredit schemes (individual versus group) and gender effects may not matter significantly, money separation does appear to have a positive and significant relationship with performance. On the policy front, this paper recommends that non-financial factors such as human capital, spousal involvement, and money separation be considered as important factors for improving microenterprise business performance in the case at least of Indonesia, with less focus on microfinance *per se*.

The rest of this paper is outlined as follows. Section 2 provides a brief literature review. Section 3 describes the research methodology. Section 4 provides empirical results with discussion. Section 5 provides policy implications. Section 6 concludes.

2. Brief literature review and hypothesis development

In this section, we develop our hypothesis with respect to each independent variable based on extant literature. With regard to those included in the Atmadja et al. (2016) study, we summarise the literature review and restate the hypotheses since we concur with them—for details, we invite the reader to peruse that study, which can be found in Atmadja et al. (2016). With regard to the three new independent variables—microcredit scheme, gender and money separation—we develop our hypothesis based on a brief review of the literature. Essentially, H₁ to H₃ below are those adopted from the Atmadja et al. (2016) study and H₄ to H₆ are developed in this study.

Hypothesis 1: The relationship between financial capital and microenterprise business performance will be negative.

Hypothesis 2: Human capital will be positively associated with microenterprise business performance.

Hypothesis 3: Social capital will be positively associated with microenterprise business performance.

2.1. Microcredit schemes and microenterprise performance

In the microfinance industry, the individual and group lending credit schemes are the most common types available to microenterprises. Under the former, the size of the loan is determined primarily on the basis of the pledged collateral, which may be repossessed in the event of default. Thus, while on a much lower scale, some parallels can be drawn between this and loans obtained from the formal sector such as banks.

Under the latter, microcredit is offered to individuals only via "groups"—an individual borrower applies for a loan through their "lending group". The lending group, assisted by an officer appointed by the microfinance provider, decides on the amount to be approved and subsequently becomes liable for repayment in the event of default². To ensure timely repayment of the loan, the group lending scheme involves frequent repayment meetings and peer pressure. For some borrowers, these might be burdensome, especially if the group is formed from communities with a high degree of social ties. Nevertheless, evidence suggests that the frequent meetings might also benefit the members. For example, using field experiments in India, Feigenberg, Field, and Pande (2010) provide evidence on the role of microfinance in building social capital by reporting that more frequent group meetings can in practice

Many microfinance providers in developing economies rely on joint liability for their business operations as a means to induce peer monitoring and reduce ex ante (Armendariz de Aghion & Morduch, 2005; Banerjee, Besley, & Guinnane, 1994; Stiglitz, 1990) as well as ex post (Besley & Coate, 1995; Bhole & Ogden, 2010) moral hazard over investment choice particularly in the absence of collateral and the providers' credit screening. However, if trust between members in a lending group is low, along with little enforcement of contracts, this may also become liabilities for the microfinance providers.

lead to greater social interactions. These interactions might provide members with alternative sources of information that might not otherwise be available to individuals, which might help discover or create new opportunities in the market (Shepherd, McMullen, & Jennings, 2007). As one of the manifestations of weak ties, a lending group can create social capital through increased communication, information diffusion, and social support (Paxton, 1999). The lending group might develop new or deepen already existing social relationships within the group that might yield economic gains and increase the likelihood of loan repayment (Anthony, 2005).

The joint liability mechanism of lending groups might also encourage risk taking behaviour by individuals (G. Fischer, 2013; Giné, Jakiela, Karlan, & Morduch, 2010). However, the strict peer monitoring practices might, on the other hand, serve to discourage such behaviour (G. Fischer, 2013).

The most recent study regarding the impact of microcredit schemes on borrowers has been reported by Attanasio et al. in 2013. Using a randomized field experiment of 1,148 women who were members of a microfinance institution (MFI) across rural Mongolia, they compared the impact of individual versus group lending microcredit on borrowers. The study finds that microcredit significantly influences business creation in the case of women receiving loans via group lending schemes. The authors argue that one possibility might be that the group lending scheme allows women to mutually insure each other against adverse business outcomes, which may in turn reduce investment uncertainty, increase loan take up, and eventually lead to larger long-run effects such as increase in business performance.

In light of the foregoing,

Hypothesis 4: Group-lending scheme funded microenterprises are likely to perform better than those funded otherwise.

2.2. Gender and microenterprise performance

Studies comparing the performance of men and women-owned firms show that those headed by women tend to be generally smaller across various dimensions, including gross revenue, number of employees, sales, assets, and profit levels (Ellis et al., 2010; E. M. Fischer, Reuber, & Dyke, 1993; Kalleberg & Leicht, 1991; Rosa, Carter, & Hamilton, 1996; Watson & Robinson, 2003). Women-owned enterprises tend to also expand more slowly than those owned by men (Cooper, Gimeno-Gascon, & Woo, 1994; GEM, 2010), which might be due to women being generally more risk-averse (Cliff, 1998) and lack relevant industry-specific experience (Loscocco, Robinson, Hall, & Allen, 1991).

Other studies report that women have narrower and more homogenous social networks³ (GEM, 2010; Loscocco, Monnat, Moore, & Lauber, 2009) and have more limited access to finance (Parker, 2009) compared to men. Yet another study Kessy (2009) finds that men-headed enterprises have higher business performance compared to those headed by women. The study also notes that men in general have higher assets, sales revenue and number of employees than women.

Although some other studies report that women had particularly lower rates of loan default compared to men (Khan, 1999; Khandker, 1998; Panjaitan-Drioadisuryo & Cloud, 1999; Remenyi, 2000)

³ Women's social networks emphasize interpersonal relationships over instrumental relationships; hence women are exposed to fewer business-relevant sources (Brush, 1992).

which might reduce financial risk to the lenders (Armendariz de Aghion & Morduch, 2005), it does not always necessarily mean that women—owned enterprises performed better than those owned by men. The low default rates could instead be due their more risk—averse nature (Cliff, 1998; Velasco & Marconi, 2004).

Women and men might become entrepreneurs for different reasons as well (Ellis et al., 2010). While men appear to be more often driven by economic opportunities ("opportunity-motivated"), women tend to be more "necessity-motivated" entrepreneurs who desire to meet the basic necessities or to just find employment. An enterprise helps women improve their earnings capabilities and at the same time enables the desired flexibility of a balanced work—family commitment (Bird & Brush, 2002; Brush, 1992; Ellis et al., 2010). Consequently, female borrowers tend to have a greater social impact compared to male borrowers because they invest more capital in family education, nutrition and health care (Armendariz de Aghion & Morduch, 2005; Blumberg, 1988; Panjaitan-Drioadisuryo & Cloud, 1999; Pitt, Khandker, & Cartwright, 2006).

In light of the foregoing,

Hypothesis 5: Men-owned microenterprises are likely to perform better than women-owned ones.

2.3. Money separation and microenterprise performance

Literature suggests that microcredit often fails to help microenterprises grow because borrowers tend to have limited skills to use the borrowed funds effectively (Adams & Von Pischke, 1992; Imai et al., 2010; Johnston & Morduch, 2008). Evidence shows that the borrowers may not use the funds for the intended purpose such as purchasing productive assets, or working capital (Barnes et al., 2001; Collins, Morduch, Rutherford, & Ruthven, 2009; De Mel, McKenzie, & Woodruff, 2008; Rutherford, 2006). There are at least two possible reasons for this inability to distinguish between intended and actual purpose of the loan—or, *money separation* as we refer to in this study. Firstly, self-control problems or lack of self-discipline can often lead individuals to neglect productive investments today that have would large payoffs in the future (Banerjee & Mullainathan, 2010; Duflo, Kremer, & Robinson, 2009). Secondly, there is a lack of access to credit services for the financially disadvantaged to smooth their daily consumption and to deal with emergency as this group are historically thought of as having no need for credit services; hence, microcredit becomes the only credit available to such families (Collins et al., 2009; Hoque, 2004).

To provide an example from Indonesia, via a survey of 1,438 households across six provinces of Indonesia in 2002, Johnston and Morduch (2008) find that at least 50% of the microcredit from MFIs to microenterprises were for purposes totally unrelated to business. Instead, the loans were used for home improvement, non-business land or building purchase, school tuition, medical treatment, other loan repayment, daily needs or retirement needs, vehicle purchase, household goods, ceremony or social expenditure, holiday needs, jewellery purchase, etc. Moreover, the authors report that female-headed households were more likely to borrow for daily consumption. The borrowers' inability to distinguish their personal/consumption expenses from business-related investment often prolongs their dependence on external finance (Parker, 2009). This lack of self-discipline is more likely to bring undesirable

consequences to the business performance of MEs. It is apparent that a good business or management practice, even in the simplest form such as separating money for business from household expenses, might be important for microbusiness success.

In light of the foregoing,

Hypothesis 6: Money separation is positively related to microenterprise business performance.

3. Research method

3.1. The survey

As indicated earlier, the present (2014) survey was conducted in the same location as the previous (2010)—in Indonesia's second largest city of Surabaya and its surroundings, from February to June 2014. The key methodological improvement to note in the 2014 compared to the 2010 surveys is in relation to the number and extent of microfinance providers covered—the 2010 survey covered only one microfinance provider Setya Bhakti Wanita (SBW); the 2014 survey covered five—four more than the previous, including two Islamic-style microfinance providers⁴ (Madani and Artha Bina Ummat), one cooperative (Assakinah) and one government-sponsored microcredit providers (BKM Bendul Merisi) plus SBW. Besides, the questionnaire has been extended to include many important questions that the earlier survey did not cover. These methodological improvements are expected to address some limitations in previous studies (Atmadja et al., 2016; Bradley et al., 2011; Neubert et al., 2017; Wood et al., 2015). Table 1 provides a summary of the features of the five providers.

Table 1: Characteristics of the microcredit providers covered in the 2014 survey

	Setya Bhakti Wanita	Artha Bina Ummat	Assakinah	Madani	BKM Bendul Merisi
Established	1978	1998	1999	2007	2010
Type	Cooperative	Islamic	Cooperative	Islamic	Community Empowerment
Total	12470 (all women)	1592 (842 men,	971 (51 men,	3515 (1439 men	205 (113 men
membership	,	750 women)	920 women)	2076 women)	92 women)
Number of lending groups	418	0	64	0	29
Credit schemes	Individual, Group lending	Individual only	Individual, Group lending	Individual only	Group lending only
Total credit outstanding (in IDR, 2013)	More than 1 billion	More than 1 billion	More than 1 billion	More than 1 billion	Less than 1 billion
Credit ranges (in IDR)	Up to 2b (individual) 1m– 27m (group)	0.5m-40m	Up to 200m (individual) 0.5m to 8m (group)	0.1m to 500m	0.5m to 4.5m
Credit terms	Up to 15 years (individual) 3 – 25 months (group)	Up to 24 months	Up to 24 months (group). 3-6 months (individual)	Up to 48 months	Up to 12 months
Credit	Collateral	Collateral	Collateral	Collateral	Voluntary

⁴ Since the microcredit providers are formally registered as multipurpose cooperatives and some of their funding sources come from commercial banks' debts which apply standard conventional banking procedures, they might not be able to strictly implement sharia principles on their lending operations.

requirements	(individual)	(individual)	saving
	Compulsory	Compulsory	
	saving and	saving and	
	voluntary saving	voluntary saving	
	(group)	(group)	

Source: the data was obtained from the 2014 survey interviews with the MFIs.

As Table 1 shows, our sample provides a reasonable mix of microcredit providers, which in turn provides a good robustness test. For example, the sample includes small, medium and large sized providers, with membership ranging from 205 to 12,470. Similarly, the sample microcredit providers are relatively new (2010) to relatively well–established (1978), cover different types—Islamic, cooperatives and others—and with different combinations of lending group versus individual credit schemes and different make up in terms of men and women memberships.

At the time of the survey, the five microcredit providers had a total membership of 17,553, of which 5,531 (205 from BKM, 3164 from SBW, 738 from Assakinah, 575 from ABU, and 849 from Madani) met the critical survey criteria of "owns a microenterprise". Of the 5,531 who meet the criteria, 1,424 (or 26%) had borrowed via individual credit schemes and the rest, 74%, had borrowed using lending group schemes.

Of the eligible respondents, those using the lending group schemes to borrow belonged to around 178 lending groups (41 from Assakinah, 108 from SBW, and 29 from BKM). From each of these groups, two to three members were randomly selected as prospective respondents—a total of 530. Respondents using the individual credit scheme were also included in the survey—around 270 were randomly selected as prospective respondents.

Thus, a total of 800 prospective respondents were identified and contacted by the sample microfinance providers on behalf of the researchers for their voluntary participation. Of these, only 556—405 belonging to group lending schemes and 151 individuals—agreed to be interviewed. Interviews were conducted mostly at the respondent's residence or business place to observe their real-life conditions; occasionally the interviews were conducted at group meetings. Of that figure, only 453 completed responses were found to be valid for the purposes of the analysis. Of the 453 respondents, 89 (20%) turned out to be men and 364 (80%) women, a ratio very much reflective of the make—up of the microfinance membership in Indonesia—there are many more women than men.

The interviewers were local university students, from a final year research methods class who voluntarily wanted to participate in the survey. Ten students were selected based on their academic qualifications and relevant prior experience. The researchers provided training to the interviewers prior to the survey, which included technical understanding about the details of the questionnaire and the implementation of the ethical conduct, and closely supervised/monitored them during the data collection process.

3.2. The variables

Dependent variable. Consistent with Atmadja et al. (2016) and for the reasons explained in that study, *profit change* is used as the proxy for measuring business performance.

Independent variables. Again, consistent with Atmadja et al. (2016) and for the reasons explained in that study, we include the following independent variables in our study: microcredit, human capital, and social capital.

In addition, as noted earlier, we include gender (1 = female; 0 otherwise), credit schemes (1 = group lending; 0 otherwise) and money separation (1 = strict separation; 0 otherwise). The term *money separation* in this study is different from the usage in Karlan and Valdivia (2010)⁵. Data was obtained by asking respondents "how do you separate your money for business and money for household expenses?" It was a close-ended question with three optional responses—no separation; not strictly; strictly. The responses were then transformed into a dummy variable (i.e. 1 for strictly, and 0 otherwise). The reason for settling with the "binary and subjective" response is that micro entrepreneurs in Indonesia tend not to keep proper records of their business transactions; quite often they are not properly trained, qualified or otherwise equipped to do so, which hindered us from obtaining more precise data on proportion of money used for business and for household expenses from the respondents, ideally required for this variable.

Control variables. The control variables are change in total assets, competition, number of employees, new products, respondents' age⁶ and length of microfinance membership.

3.3. Data description

Table 2 describes the data of selected variables, comparing as well that from the 2014 survey to the 2010 survey. In the table, data from the 2014 survey are presented in two parts – Part A and Part B.

Table 2. The descriptive statistics of the data.

Variables		2010		2014					
			Part A				Part B		
			female	male	all	female	male	all	
Profit	Decrease	18%	18%	10%	17%	16%	10%	15%	
	About the same	34%	19%	8%	18%	20%	9%	18%	
	Increase	47%	63%	82%	65%	63%	81%	67%	
Microcredit (in millions IDR)	Mean	11.35	9.98	2.03	8.80	9.66	4.36	8.62	
Education	Elementary	5%	6%	22%	8%	9%	22%	11%	
	Junior high	12%	15%	24%	16%	14%	19%	15%	
	Senior high	44%	54%	50%	54%	53%	49%	53%	
	University	38%	25%	4%	22%	24%	9%	21%	
Prior work	No	58%	83%	50%	78%	80%	58%	76%	
experience	Yes	42%	17%	50%	22%	20%	42%	24%	
Family business	No	47%	76%	72%	76%	73%	65%	72%	
background	Yes	53%	24%	28%	24%	27%	35%	28%	
Strongties	Mean	1.99	1.63	1.24	1.57	1.61	1.55	1.60	
Weak ties	Mean	7.74	1.68	1.14	1.19	1.16	1.20	1.17	
Lending group ties	Mean	31.77	25.42	5.86	22.53	25.42	5.86	22.53	
Spousal involvement	Mean	5.18	5.24	4.86	5.18	5.28	5.16	5.25	
Change in assets (%)	Mean	5.46	4.95	18.50	6.96	6.12	14.69	7.80	
Competition	No	37%	28%	44%	31%	30%	35%	31%	
	Yes	63%	72%	56%	69%	70%	65%	69%	

⁵ Karlan and Valdivia (2010) measure money separation as a binary variable equal to 1 if the MFI's client thinks that is not necessary to separate her money from that of her husband or partner or other adult in the household to control expenses and savings. It is treated as a dependent variable.

⁶ For the operational definition of the selected variables, please refer to Atmadja et al. (2016).

Number of employees	Mean	2.87	0.91	0.82	0.89	0.87	1.10	0.91
New product	Mean	1.97	3.19	2.72	3.12	3.18	3.09	3.17
Age	Mean	48.02	46.53	46.70	46.56	45.41	44.07	45.15
Gender	Male	0%	0%	100%	15%	0%	100%	20%
	Female	100%	100%	0%	85%	100%	0%	80%
Credit scheme	Individual	0%	0%	0%	0%	21%	44%	25%
	Group lending	100%	100%	100%	100%	79%	56%	75%
Length of membership	Mean	10.25	10.77	4.48	9.78	9.43	4.35	8.43
Number of obs.		130	288	50	338	364	89	453

Source: The data are summarised from the 2014 survey conducted by the researchers.

Part A is calculated based on the 2014 data of three microfinance providers offering group lending credit schemes (i.e. Assakinah, SBW, and BKM).

Part B is calculated based on the 2014 data of both credit schemes (i.e. group lending and individual) from the five microcredit providers.

Part A contains the data of three of the five sample microcredit providers offering group lending credit schemes (i.e. Assakinah, BKM, and SBW) with 338 number of observations. Data from this part are used in the analysis meant to address the first study objective, i.e. to compare our findings with that of Atmadja et al. (2016), which included group lending scheme only. Part B covers the data of both credit schemes (i.e. group lending and individual) from all of the five sample providers with 453 observations used for the analysis aimed at the second study objective, i.e. to test the influence of schemes (individual vs. group), gender and money separation on microenterprise performance.

In terms of the dependent variables, the table shows that more respondents overall in the 2014 survey experienced growth in profit and sales compared to the 2010 survey. Further, the 2014 respondents generally borrowed less than the 2010's, and men on average borrowed much less compared to women. With respect to human capital (education level, prior working experience and parents/family business background), respondents in the 2010 survey appears to better overall. Moreover, male respondents were less educated but had more family business background and prior working experience than the female respondents in the both surveys.

Although it is relatively narrower compared to the 2010 respondents, the 2014 female respondents appear to have wider social ties/networks than the male respondents. Compared to findings from other country studies (GEM, 2010; Loscocco et al., 2009), it seems that Indonesia female microentrepreneurs tend to not only involve more family members, but also more acquaintances into their networks compared to their male counterparts. The size of women's lending groups is also greater than that of men's.

A concern is also put on the mean of the weak ties as it is apparent the 2010 sample has much larger mean than the 2014's. It was suspected that this might be driven by some outliers. However, after re-estimating the model without the seemingly outliers, it was found that the result is not sensitive to those outliers.

4. Empirical results and discussion

Since the dependent variables are ordinal (i.e. 1 for decrease, 2 for about the same, and 3 for increase), the ordered probit is appropriately applied for analysis purposes. Table 3 presents the

estimation results of the models using the group lending scheme subsample. This is to address the first study objective. In the table, the estimation results of Atmadja et al. (2016) is also restated. The equation used in Atmadja et al. (2016) is applied to estimate the 2014 female sample.

Overall, when the dependent variable is profit change, both estimations yield similar results, indicating that the empirical result is not too sensitive across the two samples. Loan has a significant negative relationship with profit change in 2010 (at 10% level) and 2014 (at 5% level). Spouse involvement is the only social capital variable that matters for profit in 2014 as well as in 2010. The association of the both variables is more significant in 2014 (at 1% level) than in 2010 (at 10% level). Nevertheless, the human capital variables have indifferent results. Education significantly matters in 2010, but does not in 2014. Instead, prior work experience influences are significant to profit in 2014.

Table 3. Ordered probit estimation results of group lending samples

Variables	2010 1	2014 2		
	Coef	Coef	_	
Ein annial nanital				
Financial capital Microcredit	-0.0645**	0.0271*		
Microcredit		-0.0271*		
	(0.0275)	(0.0142)		
Human capital				
Education	0.2204*	0.0810		
	(0.1344)	(0.0987)		
Prior working experience	-0.1043	0.4278*		
	(0.2333)	(0.2517)		
Family business background	0.0755	-0.2680		
	(0.2274)	(0.1995)		
ocial capital	0.0120	0.0695		
Strong ties	-0.0138 0.0536	-0.0685		
V1-4:		(0.0627)		
Veak ties	-0.0118 0.0108	-0.0319		
anding ansum (I C tips)	0.0108	(0.0496) 0.0052		
Lending group (LG-ties)				
	(0.0146)	(0.0089)		
pousal involvement	0.1332*	0.1292***		
•	(0.0719)	(0.0500)		
usiness Control				
Change in assets	0.0254**	0.0408***		
	(0.0062)	(0.0077)		
Competition	0.4968**	-0.5134**		
	(0.2369)	(0.2010)		
Number of employees	0.0391	-0.0396		
	(0.0506)	(0.0586)		
New product	0.0706	0.0241		
	(0.1197)	(0.0479)		
ndividual control				
Age	0.0011	0.0131		
<i>U</i> -	(0.0139)	(0.0133)		
Length of membership	0.0294	-0.0068		
. <i>O</i>				

(0.0189) (0.0131)

Number of observation 130 288

Note: Dependent variable is profit change (1 if decrease, 2 if about the same, and 3 if increase) used as a proxy of business performance. Unstandardized coefficients and standard error (in parentheses) are presented.

- *** p<0.01, ** p<0.05, * p<0.1.
- Restated from Atmadja et al. (2016).
- ² Estimated using female samples of the 2014 data from three microfinance providers offering group lending credit schemes (i.e. Assakinah, SBW, and BKM).

To address the second study objective, five models are estimated using the 2014 data of the both credit schemes. In these models, while business performance remains the dependent variable, lending group ties variable is omitted from the analysis and all covariates used in Table 3 are treated as control variables. In addition, institution dummies (i.e. Arta Bina Umat, Madani, Assakinah, and BKM) and types of industry (i.e. manufacture, wholesale/retail, and services) are included as control variables in the models. The estimation results of the models are presented in Table 4.

In Model 1, microenterprise business performance is regressed on all control variables to provide a baseline model. Table 4 shows that microcredit and the other four variables (i.e. prior work experience, spousal involvement, assets change, and competition) are significantly linked to business performance. Corresponding to $H_4 - H_6$, credit scheme is then included as a covariate in Model 2, gender in Model 3, and money separation in Model 4. Finally, business performance is regressed on all variables in Model 5.

Table 4. Ordered probit estimation results of 2014 samples.

** ' 11	36 111	3.6.1.10	1110	3.6. 1.1.4	3.6. 1.1.6
Variables	Model 1	Model 2	Model 3	Model 4	Model 5
<u>Control variables</u>					
Microcredit	-0.0248**	-0.0248**	-0.0244**	-0.0257**	-0.0254**
	(0.0117)	(0.0117)	(0.0117)	(0.0118)	(0.0118)
Education	0.0659	0.0673	0.0690	0.0582	0.0639
	(0.0768)	(0.0770)	(0.0771)	(0.0771)	(0.0776)
Prior working experience	0.5140***	0.5120***	0.5240***	0.5540***	0.5590***
	(0.1830)	(0.1830)	(0.1830)	(0.1830)	(0.1830)
Family business background	-0.1950	-0.1960	-0.1930	-0.2010	-0.2010
	(0.1570)	(0.1570)	(0.1570)	(0.1570)	(0.1570)
Strong ties	-0.0399	-0.0402	-0.0451	-0.0100	-0.0151
_	(0.0563)	(0.0563)	(0.0564)	(0.0571)	(0.0573)
Weak ties	0.0032	0.0037	-0.0007	0.0020	-0.0013
	(0.0438)	(0.0438)	(0.0438)	(0.0438)	(0.0439)
Spousal involvement	0.1320***	0.1320***	0.1330***	0.1260***	0.1270***
	(0.0421)	(0.0421)	(0.0422)	(0.0424)	(0.0425)
Change in assets	0.0317***	0.0319***	0.0321***	0.0298***	0.0304***
	(0.0049)	(0.0049)	(0.0049)	(0.0049)	(0.0051)
Competition	-0.4480***	-0.4520***	-0.4560***	-0.4410***	-0.4560***
-	(0.1580)	(0.1590)	(0.1580)	(0.1600)	(0.1610)
Number of employees	-0.0206	-0.0210	-0.0326	-0.0328	-0.0455
	(0.0485)	(0.0485)	(0.0488)	(0.0489)	(0.0493)
New product	0.0428	0.0435	0.0375	0.0279	0.0230
-	(0.0394)	(0.0395)	(0.0395)	(0.0401)	(0.0404)
Age	0.0119	0.0120	0.0114	0.0119	0.0114

	(0.0100)	(0.0100)	(0.0100)	(0.0101)	(0.0101)
Length of membership	-0.0136	-0.0142	-0.0136	-0.0086	-0.0095
	(0.0117)	(0.0119)	(0.0117)	(0.0119)	(0.0122)
Arta Bina Umat	-0.1790	-0.0933	-0.3370	-0.0791	-0.0841
	(0.2260)	(0.3940)	(0.2460)	(0.2290)	(0.4120)
Madani	0.3370	0.4210	0.1430	0.4060	0.3710
	(0.2850)	(0.4280)	(0.3060)	(0.2910)	(0.4490)
Assakinah	-0.2560	-0.2670	-0.2520	-0.1990	-0.2120
	(0.2110)	(0.2140)	(0.2110)	(0.2130)	(0.2170)
BKM	-0.1980	-0.2100	-0.6830	-0.0147	-0.5200
	(0.2880)	(0.2920)	(0.4180)	(0.2940)	(0.4260)
Manufacture	0.0250	0.0268	0.0314	0.0222	0.0318
	(0.1710)	(0.1710)	(0.1710)	(0.1730)	(0.1730)
Trading	0.0201	0.0242	0.0315	-0.0208	-0.0018
	(0.1670)	(0.1670)	(0.1670)	(0.1690)	(0.1700)
Main covariates					
Lending scheme (LG=1)		0.0955			0.1710
((0.360)			(0.3650)
Gender (female=1)		(****)	-0.4780		-0.4800
()			(0.2980)		(0.3000)
Money separation			()	0.4970***	0.5020***
J - · F · · · · ·				(0.1550)	(0.1550)
				(,	(,
Number of observation	453	453	453	453	453

Note: Dependent variable is profit change (1 if decrease, 2 if about the same, and 3 if increase) used as a proxy of business performance. Unstandardized coefficients and standard error (in parentheses) are presented.

*** p<0.01, ** p<0.05, * p<0.1.

The models are estimated using the 2014 data of the both credit schemes from the five microcredit providers.

Table 4 generally show that, in all models, microcredit has negative effect on profit, while the other variables have positive impact on the business performance partially confirming hypothesis 2 and hypothesis 3. These findings are to some extent comparable with those of Atmadja et al. (2016)—and conform to the strand of literature (e.g. Adams and Von Pischke (1992), Collins et al. (2009), De Mel et al. (2008), Imai et al. (2010), Johnston and Morduch (2008)—suggesting that the results are not sensitive to sample size, and the inclusion of the three new variables (i.e. credit scheme, gender and money separation),. However, these findings are not consistent with some other studies proposing a positive impact of microcredit on business performance—see for example Awunyo-Vitor, Abankwah, & Kwansah, 2012; Barnes et al., 2001; Chen & Snodgrass, 2001; Crépon et al., 2011; Dunn & Arbuckle Jr, 2001; Kondo et al., 2008; Mahmood & Mohd Rosli, 2013; Panda, 2009; Ssendi & Anderson, 2009.

Considering entrepreneurship in Indonesia, commonly, due to a lack of options in the labour market, participation by the poor in the local market is often motivated by necessity. Because of this, they are less likely to have proper training, skills and resources necessary for conducting business (Bradley et al., 2011). Microcredit might help overcome liquidity constraints and lead to business formation (Banerjee, Duflo, Glennerster, and Kinnan (2015)). However, this potentially increases the number of imitative businesses and the degree of competition in the local market (Hannan & Freeman, 1984; Schumpeter, 1912), which might, in turn, reduce business profits.

In Model 2 and Model 5 of the tables, while the credit scheme (group vs. individual) shows a positive relationship with business performance, the association is statistically insignificant, suggesting that the type of scheme might not really matter for a microenterprise's business performance; hence rejecting hypothesis 4. This suggests that social capital created through a lending group membership may not necessarily make a significant difference to business performance. It is possibly because the agenda of group meetings have not been able to adequately capture business related matters. The group regular meetings have largely been dominated by loan repayment issues, neglecting thus more meaningful business related discussions. Our survey shows that majority (more than 70%) of the respondents with group-lending schemes ranked loan repayment as top priority when asked for the time typically spent in regular group meetings; hence, the opportunity for exchanging and gaining valuable information about business opportunities and ideas is rather limited⁷.

Secondly, in some microfinance providers, the group lending size was too large (up to 50 members per group). On the one hand, this can broaden business/social networks among the members. On the other hand, this could adversely affect the quality of peer credit screening and monitoring within groups. Particularly in the situation when numerous loan applications needed to be reviewed promptly, the group's considerations were more often based on credit ceiling and track records of the applicants with little attention paid to the purpose of the loan. This might stimulate risk taking behaviour of the applicants (G. Fischer, 2013; Giné et al., 2010), which possibly ends up with either business success or business failure. It might be true that group lending has a positive impact on business creation (Giné et al., 2010); however, it does not necessarily mean that the performance will always be better.

This study also finds that there is no significant gender effect on microenterprises business performance, as shown in Model 3 and Model 5. Although the negative sign of the variable's coefficient might indicate that men-owned microenterprises are more likely to perform better than women-owned ones, the difference is not significant statistically. This rejects hypothesis 5.

While credit schemes and gender do not significantly matter for performance, money separation apparently has an important role. The result of Model 4 and Model 5 in Table 4 show that the variable's coefficients are significant at 1% level with positive signs confirming hypothesis 6. Thus, holding other variables unchanged, microenterprises owned by micro-entrepreneurs who strictly separate their money for business from household use are more likely to experience better business performance compared to others. This finding provides an empirical support to the prior studies stating that borrowers' lack of self-discipline in using the loan could results in poor business performance and long-term dependence on external funding (Adams & Von Pischke, 1992; Imai et al., 2010; Johnston & Morduch, 2008; Parker, 2009).

With relatively higher interest rate imposed on the loan, the inability of micro-entrepreneurs to properly manage the loan might cause a financial burden to the enterprises, especially if they are not able to generate higher profits to compensate the relatively higher interest costs, leading to long-term dependence on external finance. Conversely, if the entrepreneurs are able to strictly separate the money,

The data also show that less than 5% of the group-lending respondents listed their lending group as becoming their main source of business ideas.

they might have sufficient reserves for financing working capital and/or purchasing business assets to expand their business to generate more profit. This basic business practice appears to have benefited the performance of Indonesian microenterprises.

5. Policy implications

Considering the above findings, the first implication that can be drawn is that microcredit per se might not enhance microenterprise business performance. Results using both the 2010 (Atmadja et al., 2016) and the 2014 data (this study) consistently show that microcredit negatively affects microenterprises' business performance. Parker (2009) points out that entrepreneurs are often unable to distinguish between consumption and business financing decisions. This argument is also supported by our study that respondents who strictly separated money for business and household expenses performed better in business. Hence, the lack of self-discipline or inability of micro entrepreneurs to adequately manage finances might be a key factor, suggesting the important roles of business and financial management skills for microenterprise success. To enhance business performance, micro entrepreneurs should not only be provided with easy access to finance but also be equipped with sufficient abilities to appropriately manage their business. For example, evidence from Tanzania shows that enterprises whose owners received business training performed better than those that did not (Kuzilwa, 2005). Our study highlights that training, mentorship, advisory services (in marketing, business and management), as well as networking services, should ideally complement microcredit; alternatively, as has been suggested by Karlan and Zinman (2011), lenders should carefully monitor the use of the loans. In addition, the most current study of self-employed business owners in developing countries conducted by Campos et al. (2017) suggests that a psychology-based personal initiative training approach, which inculcates a proactive mindset and focuses on entrepreneurial behaviour, might also be considered complements of microcredit in boosting microenterprise business success.

Alternatively, as has been suggested by Fafchamps, McKenzie, Quinn, and Woodruff (2011), to overcome the micro-entrepreneurs' self-control problem or lack of self-discipline, especially in utilising microcredit, microcredit might be provided 'in-kind' (e.g. purchasing equipment, inventories or raw materials), instead of in cash. This would ensure that the loan is invested in business; hence, it might help avoid using the loan for other purposes. Evidence shows a stronger impact of in-kind loans than of cash loans on business profits because more in-kind loans than cash loans end up in business (Fafchamps et al., 2011). Thus, it appears more effective for MFIs to help micro-entrepreneurs provide their business needs (i.e. equipment, inventories or raw material) by paying the money to the suppliers rather than giving cash to the entrepreneurs. Nevertheless, different cultural backgrounds and gender compositions between the Ghanaian study and this study should also be considered carefully when implementing this suggestion in Indonesia.

Regarding microcredit lending schemes, group-lending scheme might not provide any additional benefits to microentreprise business performance. Participating in a lending group, ideally, might assist its members in growing their business if activities if the group meetings can focus on business-related

conversations. This is particularly to give members opportunity to acquire information that might not be directly available to a particular individual. The information might help them not only discover or create opportunities in the market (Shepherd et al., 2007), but also to improve enterprises performance (Bruderl & Preisendorfer, 1998). Besides reducing the group size, the ability of loan officers, the representatives of microcredit providers, and group leaders could play key roles in encouraging members to have more business-related conversations.

Non-financial factors (i.e. human capital and spousal involvement) appear to be important for business success as well. Thus, it might be useful for policy makers to consider these factors as well in developing appropriate policies.

6. Conclusion

Essentially, this study attempts to mitigate the limitations of a recent study by Atmadja et al. (2016), which apparently was the first to test the effect of microcredit on the performance of microenterprises in the case of Indonesia and finds a negative relationship. The limitations of that study, based on a survey of women microenterprises undertaken in 2010 in Surabaya, Indonesia's second largest city, included relatively small sample, lacked heterogeneity, and focussed only on women. The present study is based on an extended and updated survey of microenterprises in Surabaya—with a larger, more heterogeneity sample, and includes men owned enterprises.

In addition to testing the main findings of the Atmadja et al. (2016) study, we investigate if different lending schemes (group vs. individual), gender and a micro-entrepreneurs' ability and/or willingness to separate their personal/consumption expenses from business—which we call "money separation"—might matter for business performance. Overall, we find that even with a much larger sample, including both men and women—owned microenterprises and greater heterogeneity, microfinance by itself may not necessarily improve business performance. Thus, our findings, by and large, confirm the findings of the Atmadja et al. (2016) study and conform to the strand of research which argues that microfinance may not really matter for microenterprise business performance. We also find that while microcredit schemes (individual versus group) and gender effects may not matter significantly, money separation does appear to have a positive and significant relationship with performance. Thus, on the policy front, this paper recommends that non-financial factors such as human capital, spousal involvement, and money separation be considered as important factors for improving microenterprise business performance in the case at least of Indonesia, with less focus on microfinance *per se*.

Future studies might involve an even larger, more balance and more heterogeneous sample size from across Indonesia. The inclusion of more male respondents and more variables affecting the microenterprises' business performance might be taken into consideration as well. It might also be necessary to provide a deeper explanation of the relationships noted in this study, for example, why loan size has a negative effect on profit but not on sales, why social ties/networks do not matter to performance, and whether applying alternative credit schemes (e.g. individual credit scheme) might differently affect business performance.

References

- Adams, D. W., & Von Pischke, J. D. (1992). Microenterprise Credit Programs: Déja vu. World Development, 20(10), 1463-1470.
- Anthony, D. (2005). Cooperation in Microcredit Borrowing Groups: Identity, Sanctions, and Reciprocity in the Production of Collective Goods. *American Sociological Review, 70*(3), 496-515.
- Armendariz de Aghion, B., & Morduch, J. (2005). *The Economics of Microfinance*. Cambridge, MA: The MIT Press.
- Atmadja, A., Su, J. J., & Sharma, P. (2016). Examining the Impact of Microfinance on Microenterprise Performance (Implications for Women-owned Microenterprises in Indonesia). *International Journal of Social Economics*, 43(10), 962-981.
- Awunyo-Vitor, D., Abankwah, V., & Kwansah, J. K. K. (2012). Women Participation in Microcredit and Its Impact on Income: A study of Small-scale Businesses in the Central Region of Ghana. *American Journal of Experimental Agriculture*, 2(3), 502-515.
- Banerjee, A. V., Besley, T., & Guinnane, T. W. (1994). The Neighbor's Keeper: The Design of a Credit Cooperative with Theory and a Test. *The Quarterly Journal of Economics, 109*(2), 491-515.
- Banerjee, A. V., Duflo, E., Glennerster, R., & Kinnan, C. (2015). The Miracle of Microfinance? Evidence from a Randomized Evaluation. *American Economic Journal: Applied Economics*, 7(1), 22-53.
- Banerjee, A. V., & Mullainathan, S. (2010). *The Shape of Temptation: Implications for the Economic Lives of the Poor*. NBER Working Paper No. 15973. Working Paper. National Bureau of Economic Research. Cambridge, MA.
- Barnes, C., Keogh, E., & Nemarundwe, N. (2001). Microfinance Program Clients and Impact: An Assessment of Zambuko Trust, Zimbabwe. Washington, DC: AIMS.
- Bateman, M., & Chang, H. J. (2012). Microfinance and the illusion of development: from hubris to nemesis in thirty years. *World Economic Review*, 1, 13-36.
- Besley, T., & Coate, S. (1995). Group Lending, Repayment Incentives and Social Collateral. Journal of Development Economics, 46(1), 1-18.
- Bhole, B., & Ogden, S. (2010). Group Lending and Individual Lending with Strategic Default. Journal of Development Economics, 91(2), 348-363.
- Bird, B., & Brush, C. G. (2002). A Gendered Perspective on Organizational Creation. Entrepreneurship Theory and Practice, 26(3), 41-65.
- Blumberg, R. L. (1988). Income Under Female Versus Male Control: Hypotheses from a Theory of Gender Stratification and Data from the Third World. *Journal of Family Issues*, *9*(1), 51-84.
- Bradley, S. W., McMullen, J. S., Atmadja, A. S., Simiyu, E. M., & Artz, K. (2011). Self-employed or Employed Others? Pre-Entry Capabilities, Entreprenerial Action, and the Learned Resourcefulness of Microcredit Firm Founders. *Frontiers of Entrepreneurship Research*, 31(19), 644-658.
- Bruderl, J., & Preisendorfer, P. (1998). Network Support and the Success of Newly Founded Businesses. *Small Business Economics*, 10(3), 213-225.
- Brush, C. G. (1992). Research on Women Business Owners: Past Trends, a New Perspective and Future Directions. *Entrepreneurship: Theory & Practice*, 16(4), 5-30.

- BWTP. (2013). Badan Kredit Desa (BKDs). Banking With The Poor Network. Brisbane Australia. Retrieved from
 - http://www.bwtp.org/arcm/indonesia/II_Organisations/MF_Providers/BKDs.htm
- Campos, F., Frese, M., Goldstein, M., Iacovone, L., Johnson, H. C., McKenzie, D., & Mensmann, M. (2017). Teaching personal initiative beats traditional training in boosting small business in West Africa. *Science*, *357*(6357), 1287-1290.
- Chen, M. A., & Snodgrass, D. (2001). Managing Resources, Activities, and Risk in Urban India: The Impact of SEWA Bank. Washington, DC: AIMS.
- Cliff, J. E. (1998). Does One Size Fit All? Exploring the Relationship between Attitudes towards Growth, Gender, and Business Size. *Journal of Business Venturing*, 13(6), 523-542.
- Collins, D., Morduch, J., Rutherford, S., & Ruthven, O. (2009). *Portfolios of the Poor: How the World's Poor Live on \$2 a Day*. Princeton, NJ: Princeton University Press.
- Cooper, A. C., Gimeno-Gascon, F. J., & Woo, C. Y. (1994). Initial Human and Financial Capital as Predictors of New Venture Performance. *Journal of Business Venturing*, 9(5), 371-395.
- Copestake, J., Bhalotra, S., & Johnson, S. (2001). Assessing the Impact of Microcredit: A Zambian Case Study. *Journal of Development Studies*, *37*(4), 81-100.
- Crépon, B., Devoto, F., Duflo, E., & Parienté, W. (2011). Impact of Microcredit in Rural Areas of Morocco: Evidence from a Randomized Evaluation *MIT Working Paper*. Cambridge, MA: MIT.
- Davis, M. (2006). Planet of Slums: Urban Involution and the Informal Working Class. New York: Verso.
- De Mel, S., McKenzie, D., & Woodruff, C. (2008). Returns to Capital in Microenterprises: Evidence from a Field Experiment. *The Quarterly Journal of Economics, 123*(4), 1329-1372.
- Duflo, E., Kremer, M., & Robinson, J. (2009). *Nudging Farmers to Use Fertilizer: Theory and Experimental Evidence from Kenya*. NBER Working Paper No. 15131. Working Paper. National Bureau of Economic Research. Cambridge, MA.
- Dunn, E., & Arbuckle Jr, J. G. (2001). The Impacts of Microcredit: A Case Study from Peru *USAID AIMS Project*. Washington DC: USAID Office of Microenterprise Development.
- Ellis, A. N., Orlando, M. B., Muñoz Boudet, A. M., Piras, C., Reimao, M., Cutura, J., . . . de Castro, O. (2010). Women's Economic Opportunities in the Formal Private Sector in Latin America and the Caribbean: A Focus on Entrepreneurship: Inter-American Development Bank.
- Fafchamps, M., McKenzie, D., Quinn, S., & Woodruff, C. (2011). When is Capital Enough to Get Female Microenterprises Growing? Evidence from a Randomized Experiment in Ghana. NBER Working Paper No. 17207. The National Bureau of Economic Research. Cambridge, MA.
- Feigenberg, B., Field, E. M., & Pande, R. (2010). Building Social Capital through Microfinance NBER Working Paper Series: National Bureau of Economic Research.
- Fischer, E. M., Reuber, A. R., & Dyke, L. S. (1993). A Theoretical Overview and Extension of Research on Sex, Gender, and Entrepreneurship. *Journal of Business Venturing*, 8(2), 151-168.
- Fischer, G. (2013). Contract Structure, Risk-Sharing, and Investment Choice. *Econometrica*, 81(3), 883-939.
- GEM. (2010). 2010 Report: Women Entrepreneurs Worldwide: Global Entrepreneurship Monitor.
- Giné, X., Jakiela, P., Karlan, D., & Morduch, J. (2010). Microfinance Games. *American Economic Journal: Applied Economics*, 2(3), 60-95.
- Hannan, M. T., & Freeman, J. (1984). Structural Inertia and Organizational Change. *American Sociological Review*, 49(2), 149-164.

- Holloh, D. (2001). *ProFi Microfinance Institutions Study* Retrieved from http://www.spanish.microfinancegateway.org/files/38995 file 41.pdf
- Hoque, S. (2004). Micro-credit and the Reduction of Poverty in Bangladesh. *Journal of Contemporary Asia*, 34(1), 21-32.
- Imai, K. S., Arun, T., & Annim, S. K. (2010). Microfinance and Household Poverty Reduction:

 New Evidence from India. *World Development*, *38*(12), 1760–1774.
- Johnston, D. J., & Morduch, J. (2008). The Unbanked: Evidence from Indonesia. *The World Bank Economic Review, 22*(3), 517-537.
- Kalleberg, A. L., & Leicht, K. T. (1991). Gender and Organizational Performance: Determinants of Small Business Survival and Success. *The Academy of Management Journal, 34*(1), 136-161.
- Karlan, D., & Valdivia, M. (2010). Teaching Entrepreneurship: Impact of Business Training on Microfinance Clients and Institutions. Review of Economics and Statistics, 93(2), 510-527.
- Karlan, D., & Zinman, J. (2011). Microcredit in Theory and Practice: Using Randomized Credit Scoring for Impact Evaluation. *Science*, *332*(6035), 1278-1284.
- Kessy, S. (2009). *Microfinance and Enterprises Performance in Tanzania: Does Gender Matter?*Paper presented at the A paper presented at 10th Annual IAABD Conference in.
- Khan, M. R. (1999). Microfinance, Wage Employment and Housework: A Gender Analysis. *Development in Practice*, *9*(4), 424-436.
- Khandker, S. R. (1998). Fighting Poverty with Microcredit: Experience in Bangladesh. Oxford: Oxford University Press.
- Kondo, T., Orbeta, A., Dingcong, C., & Infantado, C. (2008). Impact of Microfinance on Rural Households in the Philippines. *IDS Bulletin*, *39*(1), 51-70.
- Kuzilwa, J. A. (2005). The Role of Credit for Small Business Success: A Study of the National Entrepreneurship Development Fund in Tanzania. *Journal of Entrepreneurship*, 14(2), 131-161
- Loscocco, K. A., Monnat, S. M., Moore, G., & Lauber, K. B. (2009). Enterprising Women: A Comparison of Women's and Men's Small Business Networks. *Gender & Society, 23*(3), 388-411.
- Loscocco, K. A., Robinson, J., Hall, R. H., & Allen, J. K. (1991). Gender and Small Business Success: An Inquiry into Women's Relative Disadvantage. *Social Forces*, *70*(1), 65-65.
- Mahmood, R., & Mohd Rosli, M. (2013). Microcredit Position in Micro and Small Enterprise Performance: the Malaysian Case. *Management research review*, *36*(5), 436-453.
- Neubert, M. J., Bradley, S. W., Ardianti, R., & Simiyu, E. M. (2017). The Role of Spiritual Capital in Innovation and Performance: Evidence from Developing Economies. Entrepreneurship Theory and Practice, 41(4), 621-640.
- Olson, P. D., Zuiker, V. S., Danes, S. M., Stafford, K., Heck, R. K. Z., & Duncan, K. A. (2003). The Impact of the Family and the Business on Family Business Sustainability. *Journal of Business Venturing*, *18*(5), 639-666.
- Panda, K. (2009). Participation in the Group Based Microfinance and its Impact on Rural Households: A Quasi-experimental Evidence from an Indian State. *Global Journal of Finance and Management*, 1(2), 171-183.
- Panjaitan-Drioadisuryo, R. D. M., & Cloud, K. (1999). Gender, Self-employment and Microcredit Programs: An Indonesian Case Study. *The Quarterly Review of Economics and Finance*, 39(5), 769-779.
- Parker, S. C. (2009). The Economics of Entrepreneurship. New York: Cambridge University Press.
- Paxton, P. (1999). Is Social Capital Declining in the United States? A Multiple Indicator Assessment. *American Journal of Sociology*, *105*(1), 88-127.

- Pitt, M. M., Khandker, S. R., & Cartwright, J. (2006). Empowering Women with Micro Finance: Evidence from Bangladesh. Economic Development and Cultural Change, 54(4), 791-831.
- Remenyi, J. (2000). Is There a 'State of the Art' in Microfinance? In J. Remenyi & B. Quinoñes (Eds.), Microfinance and Poverty Alleviation: Case Studies from Asia and the Pacific (pp. 25-61). London: Pinter.
- Rosa, P., Carter, S., & Hamilton, D. (1996). Gender as a Determinant of Small Business Performance: Insights from a British Study. Small Business Economics, 8(6), 463-478.
- Rutherford, S. (2006). Uses and Users of MFI Loans in Bangladesh. MicroSave. Briefing Notes on Grameen II. MicroSave.
- Schumpeter, J. A. (1912). A Theorie der Wirt-schaftlichen Entwicklung (The Theory of Economic Development) (R. Opie, Trans.). Cambridge, MA: Harvard University Press.
- Shepherd, D. A., McMullen, J. S., & Jennings, P. D. (2007). The Formation of Opportunity Beliefs: Overcoming Ignorance and Reducing Doubt. Strategic Entrepreneurship Journal, 1(1-2), 75-95.
- Ssendi, L., & Anderson, A. R. (2009). Tanzanian Micro Enterprises and Micro Finance: The Role and Impact for Poor Rural Women. Journal of Entrepreneurship, 18(1), 1-19.
- Stiglitz, J. E. (1990). Peer Monitoring and Credit Markets. The World Bank Economic Review, (3), 351-366.
- Velasco, C., & Marconi, R. (2004). Group Dynamics, Gender and Microfinance in Bolivia. Journal of International Development, 16(3), 519-528.
- Watson, J., & Robinson, S. (2003). Adjusting for Risk in Comparing the Performances of Maleand Female-controlled SMEs. Journal of Business Venturing, 18(6), 773-788.
- in Con,
 usiness Vem.
 ots, Reasons, an
 ice Economies. Journ. Wood, M. S., Bradley, S. W., & Artz, K. (2015). Roots, Reasons, and Resources: Situated Optimism and Firm Growth in Subsistence Economies. Journal of Business Research, (1), 127-136.

Surja Atmadja¹, Iv. dent of Economics, University of Accounting Finance and Eu ang author: p.sharma@griffith.edu.au Microfinance and microenterprise performance in Indonesia: an extended and updated survey

Bukti konfirmasi artikel accepted (27 Oktober 2017)

Re: International Journal of Social Economics - Decision on Manuscript ID IJSE-02-2017-0031.R1

Parmendra Sharma <p.sharma@griffith.edu.au>

Min 29/10/2017 11.30

Kepada: Adwin Surja A. <aplin@petra.ac.id>

Cc: Jen-Je Su <j.su@griffith.edu.au>;adwin.atmadja@griffithuni.edu.au <adwin.atmadja@griffithuni.edu.au> Congrats, Adwin! You now have another pub. Let's now work on the SWB paper and try to submit to at least an A.

cheers, PS

On 29 October 2017 at 10:34, Adwin Surja A. aplin@petra.ac.id> wrote:

Thanks, Parmendra & JJ.

Congratulation.

On Sat, 28 Oct 2017 at 8:16 am, Jen-Je Su < j.su@griffith.edu.au > wrote:

Thanks, Parmendra. Congratulations, Adwin and Parmendra! JJ

On Sat, 28 Oct 2017 at 6:18 am, Parmendra Sharma < p.sharma@griffith.edu.au > wrote:

Dear Professor Connelly,

Thank you very much for this great news--we will certainly do as requested. We look forward to seeing the paper in print soon.

Kind regards,

Parmendra

On Saturday, 28 October 2017, International Journal of Social Economics < onbehalfof+J.Connelly+hull.ac.uk@manuscriptcentral.com wrote:

27-Oct-2017

Dear Dr. Sharma:

It is a pleasure to accept your manuscript entitled "Microfinance and microenterprise performance in Indonesia: an extended and updated survey" in its current form for publication in International Journal of Social Economics. The comments of the reviewer(s) who reviewed your manuscript are included at the foot of this letter.

By publishing in this journal, your work will benefit from Emerald EarlyCite. This is a pre-publication service which allows your paper to be published online earlier, and so read by users and, potentially, cited earlier.

Please go to your Author Centre at https://mc.manuscriptcentral.com/ijsec (Manuscripts with Decisions for the submitting author or Manuscripts I have co-authored for all listed co-authors) to complete the copyright assignment form. We cannot publish your paper without this. All authors are requested to complete the form and to input their full contact details. If any of the contact information is incorrect you can update it by clicking on your name at the top right of the screen. Please note that this must be done prior to you submitting your copyright form. If you would like more information about Emerald's copyright policy, please visit the Information & Forms section in your Author Centre.

If you have an ORCID please check your account details to ensure that your ORCID is validated.

You may submit or include the metadata only of your accepted article, and provide the DOI and link to the Version of Record (i.e. the official PDF of your published paper), to SSRN, your ResearchGate, Academia.edu, or Mendeley profile (or any other commercial repository). For more information about Emerald author rights, please visit: http://www.emeraldgrouppublishing.com/authors/writing/author_rights.htm

FOR OPEN ACCESS AUTHORS: Please note if you have indicated that you would like to publish your article as Open Access via Emerald's Gold Open Access route, you are required to complete a Creative Commons Attribution Licence - CCBY 4.0 (in place of the standard copyright assignment form referenced above). You will receive a follow up email within the next 30 days with a link to the CCBY licence and information regarding payment of the Article Processing Charge. If you have indicated that you might be eligible for a prepaid APC voucher, you will also be informed at this point if a voucher is available to you (for more information on APC vouchers please see http://www.emeraldpublishing.com/oapartnerships

Thank you for your contribution. On behalf of the Editors of International Journal of Social Economics, we look forward to your continued contributions to the Journal.

Sincerely, Prof. James Connelly Editor, International Journal of Social Economics J.Connelly@hull.ac.uk

Reviewer(s)' Comments to Author:

--

Sent from Gmail Mobile

--

Jen-Je Su, PhD (Michigan State Univ)
Senior Lecturer, HDR Convenor (Economics)
Department of Accounting, Finance & Economics
Griffith University (Nathan Campus)
Brisbane, Queensland
Australia

Adwin S Atmadja, PhD (Griffith Univ) Department of Accounting Faculty of Business & Economics Petra Christian University Surabaya-Indonesia

email: aplin@petra.ac.id